



A Review on State-of-Charge Estimation Methods, Energy Storage Technologies and State-of-the-Art Simulators: Recent Developments and Challenges

Tawanda Kunatsa * D, Herman C. Myburgh D and Allan De Freitas D

Department of Electrical, Electronic and Computer Engineering, University of Pretoria, Pretoria 0028, South Africa; herman.myburgh@up.ac.za (H.C.M.); allan.defreitas@up.ac.za (A.D.F.) * Correspondence: tawanda.kunatsa@tuks.co.za

Abstract: Exact state-of-charge estimation is necessary for every application related to energy storage systems to protect the battery from deep discharging and overcharging. This leads to an improvement in discharge efficiency and extends the battery lifecycle. Batteries are a main source of energy and are usually monitored by management systems to achieve optimal use and protection. Coming up with effective methods for battery management systems that can adequately estimate the state-of-charge of batteries has become a great challenge that has been studied in the literature for some time. Hence, this paper analyses the different energy storage technologies, highlighting their merits and demerits. The various estimation methods for state-of-charge are discussed, and their merits and demerits are compared, while possible applications are pointed out. Furthermore, factors affecting the battery stateof-charge and approaches to managing the same are discussed and analysed. The different modelling tools used to carry out simulations for energy storage experiments are analysed and discussed. Additionally, a quantitative comparison of different technical and economic modelling simulators for energy storage applications is presented. Previous research works have been found to lack accuracy under varying conditions and ageing effects; as such, integrating hybrid approaches for enhanced accuracy in state-of-charge estimations is advised. With regards to energy storage technologies, exploring alternative materials for improved energy density, safety and sustainability exists as a huge research gap. The development of effective battery management systems for optimisation and control is yet to be fully exploited. When it comes to state-of-the-art simulators, integrating multiscale models for comprehensive understanding is of utmost importance. Enhancing adaptability across diverse battery chemistries and rigorous validation with real-world data is essential. To sum up the paper, future research directions and a conclusion are given.

Keywords: state-of-charge; battery management system; battery energy storage technologies; modelling and simulation tools; state-of-charge estimation methods

1. Introduction

Researchers in academia and industry are placing a high priority on designing energy storage systems as a result of the recent global movement to replace fossil fuels with renewable energy sources for power generation [1]. Battery management systems (BMSs) have drawn more attention in this area over other energy storage technologies owing to their lower self-discharging and excellent power densities. A BMS provides accurate information on the battery's condition and ensures that the energy within the battery is used safely and effectively [2]. The block diagram of a BMS, as shown in Figure 1, consists of various sensors, controllers, actuators and signal lines.

The BMS has the capacity to signal if the battery is operated in an abnormal situation. This is attained by controlling and monitoring the working process of the battery [3]. The control circuit evaluates the battery's state-of-charge (SoC), state-of-health (SoH), state-of-available-power-capability (SoP) and state-of-life (SoL) by employing methods that involve



Citation: Kunatsa, T.; Myburgh, H.C.; De Freitas, A. A Review on State-of-Charge Estimation Methods, Energy Storage Technologies and State-of-the-Art Simulators: Recent Developments and Challenges. *World Electr. Veh. J.* 2024, *15*, 381. https:// doi.org/10.3390/wevj15090381

Academic Editor: Michael Fowler

Received: 4 July 2024 Revised: 12 August 2024 Accepted: 20 August 2024 Published: 23 August 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).





Figure 1. Block diagram of a battery management system.

This study mainly focuses on the SoC since it is an important parameter for most systems that require batteries and also a lingering issue that is yet to be fully explored in research and development [5]. The SoC is seen as a crucial index utilised in accessing photovoltaic (PV) battery storage systems for home energy management systems (HEMSs) [6]. In this case, the effective battery SoC information should be availed to the user to either switch off or reduce the number of electrical appliances within the home to save more energy. A battery's SoC is the proportion of its total capacity that remains untapped at any given moment [7]. It can also be described as the battery's charge level compared to its full capacity, where 0% represents empty and 100% represents a full charge [8]. Furthermore, the SoC is seen as a crucial state that needs to be observed in order to enhance performance and increase system longevity [4]. Generally, the SoC is defined by the ratio of the system's current capacity (Q_t) to its nominal capacity (Q_n) as follows:

$$oC_t = \frac{Q_t}{Q_n} \tag{1}$$

Knowing the exact SoC not only gives details about the real-time energy status of batteries but also provides assurance of smooth operation [9]. The accuracy of SoC estimation is a challenge since batteries are electrochemical devices that are complex, with unique characteristics that depend on their different internal and external conditions [10]. Furthermore, factors that affect battery performance, for instance, temperature, current rate, ageing of the battery and measurement noise, make accurate estimation of the SoC difficult. Hence, researchers have endeavoured to come up with effective SoC estimation approaches for batteries [11]. In the work by Hannan et al. [6], they suggested employing machine learning methods—specifically, recurrent nonlinear auto-regressive models with exogenous

S

inputs and the lightning search algorithm (LSA)—to improve the accuracy, efficiency and speed of SoC estimation. They confirmed the effectiveness of their method by conducting experiments on lithium–ion batteries, accounting for temperature, ageing effects and noise, demonstrating its superiority over alternative approaches. The authors asserted that their method is ideal for online battery management systems due to its rapid real-time SoC estimation, which is facilitated by minimal computational complexities during testing.

The authors in [12] devised a battery monitoring system using supervised machine learning to gauge both the battery's health and its SoC. This model was crafted to forecast the battery's behaviour in real-time, furnishing users with insights into its performance. Hemavathi [13] introduced and compared the use of feedforward neural networks and layer recurrent neural networks employing scaled conjugate gradient (SCG) and Levenberg-Marquardt (LM) training methods for estimating the battery SoC. Conducted with an 18,650 single-cell Li-ion battery in real-time using MATLAB, the experiment revealed that the LM method surpassed others for accurately estimating the Li-ion battery parameters such as the SoC, Narayanan et al. [14] suggested a machine-learning-based approach that links the battery static SoC and open-circuit voltage (OCV) utilising real-time data from lithium–ion batteries at various temperatures. Their experiments showcased the neural network approach's superior performance, displaying high R-square and low root mean square error (RMSE) indices compared to other techniques.

Lithium–ion batteries are essentially widely used because of their great energy density, extended lifespan and environmentally friendly characteristics. However, irrespective of these factors, their capacity begins to drop gradually (70–80%) after a number of years when used in EVs. This affects the energy and power capacity requirements, and as time goes on, such batteries must be retired or withdrawn from use to maintain safe operations and the provision of adequate power supply [15,16]. Research has shown that such batteries that are retired from EV applications can still serve a useful life in applications such as energy storage plants, electric bicycles and smart grids [17,18]. Hence, delving into the efficiency and dependability of repurposed retired lithium–ion power batteries is essential. This research will further support and foster the widespread adoption and sustainable growth of electric vehicles (EVs).

In the existing literature [19], retired lithium–ion batteries face inconsistencies stemming from their initial manufacturing, making their utilisation in secondary lifecycles challenging. These batteries undergo diverse usage scenarios leading to performance degradation, susceptibility to overcharging or over-discharging and potential risks such as ageing, thermal runaway and hazards like explosions [20,21]. In [22,23], the authors stress the significance of assessing and screening retired lithium–ion batteries before integrating them into battery packs, aiming to preserve their remaining capacity and ensure ongoing safe operation of the battery system. It is essential to acknowledge that despite these conditions, retired lithium–ion batteries cannot match the precision of newly manufactured ones. Their distinct parameters: terminal voltage, open-circuit voltage, internal resistance, charge/discharge current and SoC exhibit notable inconsistencies [23]. Therefore, achieving accurate estimates of the remaining capacity of retired batteries is vital for sustaining the power needs, durability and safety of the battery pack.

In Loukil et al.'s study [4], a comprehensive review was proposed to determine battery parameters, particularly focusing on the SoC. The methods used in this evaluation combined several strategies, making it difficult to determine a direct correlation between method robustness and accuracy. However, these techniques kept their execution complexity modest [24]. Dao et al. [25] introduced a lithium–ion battery SoC estimation design using a combination of a Kalman filter and artificial neural network. This approach, trained with Google TensorFlow's open-source library, exhibited superior performance, displaying an error rate of less than 1% compared to other methodologies. Additionally, Dewalkar et al. [26] demonstrated the effectiveness of artificial intelligence methodologies over traditional approaches, leveraging realistic battery testing for training purposes. Table 1 consolidates the diverse approaches utilised by researchers for developing SoC estimation methods for batteries. Despite extensive research in this domain, accurate estimation of battery SoC remains a persistent challenge. Hence, this article presents a survey focused on SoC estimation methods tailored for energy storage systems, particularly in the context of photovoltaic system applications. The contributions of this work are as follows:

- SoC estimation methods for batteries are discussed, and much focus is given to the different methods applicable. Specific application approaches in the literature are presented based on different SoC methods.
- A battery energy storage system is a major requirement for a reliable and sustainable power supply. From this perspective, commonly used battery technologies for solar PV configurations and their unique characteristics are presented.
- An overview and a quantitative comparison of the technical and economic modelling simulators for energy storage applications are presented in order to assist researchers in decision-making on the choice of potential simulators. Summaries of the potential modelling simulators for designing, testing and analysing battery energy storage are given.
- The intermittent nature of renewable energy sources is highlighted, and recommendations for novel approaches for battery SoC estimations in solar PV applications for sustainable and reliable power supplies in austere and remote communities are given.

The subsequent sections of this article are outlined as follows: Section 2 presents a discussion of energy storage technologies with an emphasis on battery energy storage systems and their characteristics, and this is followed by parameters for deep-cycle batteries for PV applications. Section 3 discusses the different SoC estimation methods, including direct-measurement-based methods, bookkeeping-based methods and adaptive-based methods. Additionally, the advantages and disadvantages of the various estimation techniques are discussed. Section 4 provides an outline and quantitative comparison of both the technical and economic modelling instruments designed for applications in energy storage. Ultimately, Section 5 gives the paper's conclusion and points out some recommendations and possible future research directions.

Table 1. Summary of research works on SoC estimation for battery storage systems.

Authors	Year	Remarks
Hannan et al. [6]	2020	Recurrent nonlinear auto-regressive techniques paired with the lightning search algorithm (LSA) were employed to improve SoC estimation, aiming for higher accuracy and performance and faster convergence. Validation of their approach was conducted through experiments involving lithium–ion batteries, considering factors like temperature variations, ageing effects and noise interference.
Varshney et al. [12]	2021	A monitoring system for batteries was created utilising supervised machine learning to evaluate both the battery's health and its SoC. This model can forecast the real-time behaviour of the battery.
Hemavathi et al. [13]	2010	Evaluated the estimation of batteries' SoC by contrasting the performance of feedforward neural networks and layered recurrent neural networks, employing scaled conjugate gradient (SCG) and Levenberg–Marquardt (LM) training methods.
Narayanan et al. [14]	2022	Suggested a machine-learning-driven approach for determining battery static SoC through open- circuit voltage (OCV) analysis. The method's effectiveness was assessed and compared using real-time lithium–ion battery data collected at various temperatures.
Loukil et al. [4]	2021	Proposed a comprehensive analysis aiming to identify battery parameters, particularly focusing on SoC. The techniques employed encompassed a blend of diverse approaches.
Dao et al. [25]	2021	Introduced a design for estimating the SoC of lithium–ion batteries. Their method integrated a Kalman filter and artificial neural network approach, constructed specifically for SoC estimation and trained using the open-source Google TensorFlow library.
Dewalkar et al. [26]	2022	Showed how artificial intelligence approaches are more effective compared to conventional meth- ods since they are trained to utilise realistic tests of batteries.

Energy storage technologies can be grouped into various types depending on their utilisation, such as thermal, flywheel, thermochemical, pumped, compressed air, magnetic, electrochemical/battery and chemical. For the interest of this work, discussions in this section are based on the different and available battery energy storage that can be utilised for general applications, buildings and communities, transportation, energy utilities and renewable energy utilisation. For evaluation, the main components and related characteristics are presented to indicate the difference between the various battery energy storage strategies.

Battery energy storage technology has undergone substantial advancements, becoming a pivotal solution in the quest for efficient and sustainable energy systems. The evolution of battery chemistries, especially lithium–ion batteries, has significantly improved energy density and cycle life, enabling their widespread deployment in various applications, from portable electronics to electric vehicles and grid-scale storage. However, despite these strides, critical research gaps persist. One notable gap lies in the quest for alternative materials beyond lithium–ion, aiming to enhance energy density, safety and sustainability. Exploring and developing novel chemistries such as solid-state batteries, metal-air batteries, or nextgeneration lithium-based chemistries remains a focal point. Challenges abound, including the need for scalable production methods, the understanding of complex electrolyte interactions, and ensuring cost-effectiveness without compromising performance or safety [27,28].

Moreover, the development of effective management systems to optimise battery performance, enhance safety and extend the lifespan presents another significant research gap. Intelligent battery management systems (BMSs) are crucial for real-time monitoring, accurate state estimation and implementing appropriate control strategies to maximise efficiency and durability. Addressing these gaps in BMSs involves integrating advanced algorithms for state-of-charge and state-of-health estimation, mitigating issues related to cell imbalances and thermal management, and devising smarter control strategies for diverse applications. Additionally, ensuring the cybersecurity of these systems is becoming increasingly imperative, demanding robust measures to protect against potential cyber threats [29–31]. Closing these research gaps is crucial for advancing battery technology towards safer, more efficient and more sustainable energy storage solutions for the future.

2.1. Electrochemical and Battery Energy Storage

Photovoltaic (PV) solar systems serve as an energy generation source, allowing storage through a battery storage system (BSS). This ensures a relatively steady energy supply for later use, addressing the fluctuating power output of PV systems caused by the intermittent nature of renewable energy sources. Batteries are one of the available energy storage technologies that are commonly used in recent times. A battery energy storage system (BESS) represents cutting-edge technology designed to store electrical energy for various applications within power systems. A BESS solution is based on the combination of different low-voltage power battery cells that are connected either in series or parallel to produce the required electrical capacity. It has been shown that a BESS is a combination of a control and power-conditioning system (C-PCS), batteries and other related plant accessories for reliability and sustainability of the entire system [32]. Electricity can be stored electrochemically in batteries and capacitors. These batteries are effective energy storage systems that are known for their high voltages and energy densities. Examples of such batteries encompass lithium-ion (Li-ion), nickel-cadmium (NiCd), lead-acid (PbA), sodium-sulfur (NaS), lead-carbon and flow batteries [33-35]. Capacitors such as electrochemical capacitors, electrostatic capacitors and electrolytic capacitors can also be used for the storage and delivery of energy electrochemically.

2.1.1. Lead-Acid (PbA) Battery

Lead-acid batteries are preferable for use in PV configurations considering their availability in terms of the base cost-per-watt, reliability and good energy storage density. Lead battery cells consist of two lead plates submerged in a diluted sulfuric acid solution, which generates approximately 2 V between the plates. To produce 12 V batteries, these cells are connected in series [36]. The grid structure within a typical lead-acid battery is composed of a lead alloy. As pure lead lacks structural strength, additional metals like calcium, antimony, tin and selenium are commonly added as fortifying agents to the alloy. Based on these additives, they are commonly called "lead-calcium" and "lead-antimony" batteries [37]. The charging process of lead–acid is not fast, and for most types, it takes up to 14–16 h for a full charge. Additionally, it is crucial to emphasise that the battery should be stored at its maximum SoC. Allowing the charge to drop too low can lead to sulfation, a condition that significantly impacts the battery's performance [38]. Lead–acid has the best charge retention among rechargeable batteries, has a moderate lifecycle and is not subject to memory as compared to nickel-based systems [37].

Various configurations of lead–acid batteries are accessible, including flooded or ventbased lead–acid and sealed or maintenance-free lead–acid. Flooded lead–acid batteries are equipped with vents to allow hydrogen gas, produced during electrolysis, to escape. Consequently, close attention is needed to monitor the electrolyte level and replenish it using distilled water. The initial sealed lead–acid batteries emerged in the mid-1970s [39], sparking debate among engineers regarding the term "sealed lead–acid" as misleading due to the inability to completely seal lead–acid batteries. To manage venting during highstress charging or rapid discharging, mechanisms to alleviate pressure buildup have been integrated. These batteries contain less electrolyte (known as "acid-starved") compared to flooded lead–acid types [37]. The prevalent varieties of sealed lead–acid batteries are gel, also referred to as valve-regulated lead–acid (VRLA), and absorbent glass mat (AGM). It is also worth noting that smaller packs having capacities of up to 30 Ah are referred to as SLA (sealed lead–acid) [37].

Lead–acid batteries are known for their affordability and reliability but have a limited depth of discharge (DoD). To maximise their lifespan, it is recommended to limit the DoD to around 50% [40]. Discharging beyond this threshold can drastically reduce the number of charge cycles, often leading to a lifespan reduction by up to 50% if regularly discharged to 80% DoD [41]. Optimal charging rates for lead–acid batteries range between C/10 to C/20, with a nominal voltage of about 2.23–2.25 V per cell for float charging [42,43]. Overcharging or maintaining a high SoC beyond the recommended voltage can lead to sulfation, decreasing both capacity and efficiency. Properly managing these parameters, such as avoiding deep discharging and overcharging, is crucial to enhancing the battery's efficiency and extending its service life.

In summary, these batteries are generally cost-effective and easy to manufacture, have adequate performance in low and high temperatures and have high specific power that can result in high discharge currents. However, irrespective of the above merits, some of their limitations are that they take a longer time to be charged and their lifecycle is limited. The battery lifespan reduces due to repeated deep-cycling, they cannot be stored in undercharged conditions due to sulfation, and finally, the flooded type requires watering. Table 2 presents some of the applications of lead–acid batteries.

Lead-Acid Battery	Applications				
Sealed lead–acid (SLA)	These batteries are suitable for wheelchairs, lighting emergencies and small UPS sys- tems [37]. SLAs are the best option for healthcare usage in hospitals and retirement communities because of their low cost, dependable service and low maintenance re- quirements.				
Absorbent glass mat (AGM)	These batteries can be used in marine and recreational vehicles (RVs) as well as for starter batteries for motorbikes and micro-hybrid cars.				
Valve-regulated lead-acid (VRLA)	These batteries are used as a backup power source for numerous locations, including banks, hospitals, Internet hubs and cellular booster towers.				

Table 2. Applications of lead-acid batteries.

2.1.2. Nickel-Cadmium Battery (NiCd)

Because of its many advantages over existing lead–acid batteries such as long lifespan, resilience to excessive discharges, low maintenance requirements, low-temperature capacity retention and non-critical voltage regulation, a NiCd battery is better suited for use in stand-alone photovoltaic systems. The cell is submerged in an alkaline potassium hydroxide (KOH) electrolyte solution and comprises positive electrodes made of nickel hydroxide (NiOH2) and negative electrodes composed of cadmium (Cd). These batteries are very costly but are more convenient for use when dealing with harsh weather conditions. The airline industry widely accepts the standard nickel–cadmium battery, which is regarded as one of the most resilient batteries [44]. However, its longevity requires careful consideration [44]. Nickel–cadmium and its classes, including nickel-metal-hydride (NiMH), possess a memory effect that, if ignored for an entire periodic discharge cycle, causes capacity loss.

NiCd batteries are resilient to deep discharges and can be fully discharged to 100% DoD without significant degradation, making them suitable for applications requiring high discharge rates [45]. However, they suffer from the memory effect, which necessitates periodic full discharge cycles to maintain optimal capacity. The ideal charging rate for NiCd batteries is typically between C/10 and C/2, with a nominal charging voltage of about 1.4 V per cell [46]. Consistent overcharging or improper charging can lead to increased internal resistance and capacity loss, significantly reducing the battery's longevity. Optimizing the charge–discharge cycles by adhering to the recommended rates can improve both the efficiency and the lifecycle of NiCd batteries, especially in high-demand applications.

In summary, NiCd batteries can easily be fast-charged with little stress, they have a long shelf life and can be stored in a discharged state, which requires priming before use. They have a good load and low-temperature performance and are rugged batteries with high cycle counts and adequate maintenance. Some of their limitations include but are not limited to the fact that they have a memory effect, they are are subjected to high self-discharge, cadmium is toxic, they are not ideal for disposal in landfills, and they have a low cell voltage of 1.20 V (which requires more cells to produce high voltage).

2.1.3. Sodium–Sulfur Battery (NaS)

The NaS battery is one of the emerging alternatives for high-power energy storage systems. NaS is a molten metal battery that is manufactured based on sodium as the anode and sulfur as the cathode. This technology uses ceramic Beta-*AI*₂*O*₃ as a form of electrolyte and separator at the same time [47]. The chemistry of the reaction between ions transfers the sodium ions to the cathode, which react with sulfur anions to produce sodium polysulphide NaSx. NaS batteries are well-known for their extremely high energy density, excellent charge and discharge efficiency (89–92%), extended lifespan and easily accessible materials [48]. These cells are mostly utilised for large-scale non-mobile applications due to their high temperature of operation, which ranges from 300 °C to 350 °C [49] and the high reactivity of the electrodes [35,48].

Sodium–sulfur batteries operate at high temperatures (300–350 °C), making them suitable for large-scale energy storage systems [50]. They typically function efficiently at a DoD of 80–90%, with an optimal charge rate around C/10 [51]. Discharging beyond this range can lead to rapid capacity degradation due to the high reactivity of sodium and sulfur at elevated temperatures. The charge voltage for NaS batteries generally ranges between 1.78–2.0 V per cell [52]. Precise control of the charge–discharge cycle and maintaining the operating temperature within the ideal range are critical to ensuring the battery's longevity and performance. Studies indicate that maintaining these parameters can significantly extend the battery's lifecycle and improve its energy efficiency.

The pros and cons of sodium–sulfur batteries are as presented in Table 3.

Pros	Cons
 Raw materials are readily available and there is no maintenance configuration Possesses high cell voltage and excellent energy and power density Energy efficiency is high High cycle life; liquid electrodes Lack of responsiveness to surrounding environments; enclosed systems operating at elevated temperatures SoC voltage rise; top-of-charge and end-of-discharge determination 	 Highly corrosive nature of the electrodes, which has high combustion with water Its temperature range of operation is within 300–350 °C It requires strict operation and maintenance It requires additional cost for the enclosed structure to avoid leakage

Table 3. Summary of the advantages and limitations of sodium-sulfur batteries, adapted from [48].

2.1.4. Lithium–Ion Battery (Li-Ion)

Lithium–ion (Li-ion) batteries commonly find application in electronic devices, including laptops, cameras, mobile phones and electric vehicles, among others. Li-ion battery cells consist of anolytic and catholytic plates and are immersed in a liquid electrolyte solution [53]. The catholyticplate is made of lithium cobaltate ($LiCoO_2$), and the anolytic plate is based on porous carbon. The electrolyte is a non-aqueous organic solution dissolved in lithium salts [35]. The flow of ions takes place from the anode to the cathode via the electrolyte and separator during the discharge process, and charging is the reverse process. This battery type has the highest energy density, is known to be safe and has no memory or cycling requirements to prolong the lifespan [54]. Li-ion batteries are of different varieties but all have a unique feature: the "lithium–ion". Their performance varies, with unique characteristics based on the choice of the active materials used. Examples of such batteries include lithium cobalt oxide (LCO), lithium manganese oxide (LMO), lithium nickel manganese cobalt oxide (NMC), lithium iron phosphate (LFP), lithium nickel cobalt aluminum oxide (NCA) and lithium titanate (LTO).

Recent advancements in lithium battery technology have introduced groundbreaking methodologies and designs that significantly enhance battery performance and safety. One of the most notable innovations is the development of solid-state batteries, which replace the conventional liquid electrolyte with a solid electrolyte [55]. This design minimises the risk of leakage and combustion, thereby enhancing safety. Moreover, solid-state batteries offer higher energy densities, which translate to longer battery life and improved performance [56]. The research by Manthiram et al. highlights that solid-state electrolytes can effectively mitigate dendrite formation: a common issue in lithium–ion batteries that can lead to short circuits and potential battery failure [57].

Another significant advancement is the implementation of nanotechnology in lithium battery design. Nanomaterials such as nanostructured silicon anodes have shown remarkable potential for increasing the capacity and efficiency of batteries [58]. Silicon anodes can store significantly more lithium ions compared to traditional graphite anodes, leading to higher energy densities [59]. However, silicon expands and contracts significantly during charge–discharge cycles, which can cause mechanical degradation. Researchers have addressed this challenge by designing silicon nanowires and nanoparticles that can accommodate these volume changes without compromising structural integrity, thus improving battery lifespan and performance [58,60]. Additionally, the use of advanced coating techniques and novel electrode architectures has contributed to enhanced thermal stability and reduced degradation rates, further improving the safety and longevity of lithium batteries [61].

The integration of machine learning and artificial intelligence in battery management systems is also a groundbreaking development [62]. These technologies enable real-time monitoring and predictive maintenance, which enhance the overall performance and safety of lithium batteries [63]. By analysing large datasets of battery usage, machine learning algorithms can predict potential failures and optimise charging cycles to extend battery life [64]. This proactive approach to battery management significantly reduces the risk of overheating and other safety hazards [65].

Lithium–ion batteries are widely used due to their high energy density and efficiency. The optimal depth of discharge for Li-ion batteries is about 80%, balancing between capacity and lifespan [66]. Discharging beyond this point, especially below 20% SoC, can significantly reduce cycle life, potentially dropping to as low as 300–500 cycles at 100% DoD compared to over 2000 cycles at 80% DoD [41,67]. The optimal charge rate is typically 0.5C to 1C, with a nominal voltage of around 3.6–3.7 V per cell, although fast charging at higher rates can increase the risk of thermal runaway if not properly managed [68]. Maintaining the charge–discharge cycles within these parameters, particularly avoiding deep discharges and high charge rates, can greatly enhance both the efficiency and longevity of Li-ion batteries, making them ideal for high-performance applications such as electric vehicles and portable electronics.

In summary, lithium–ion batteries have high specific energy and load capacities due to their power cells. They have low self-discharge, have a simple charging system with short charge times and have high capacity. They have low internal resistance and good coulumbic efficiency. A few drawbacks involve requiring a protective circuit to avert overheating under stress and facing considerable difficulty in achieving fast charging in low temperatures. Recent innovations in lithium battery technology, including solid-state designs, nanotechnology applications and advanced battery management systems, have led to substantial improvements in performance and safety. These advancements address critical issues such as energy density, thermal stability and mechanical degradation, paving the way for safer and more efficient energy storage solutions. The incorporation of solidstate electrolytes, nanostructured materials and intelligent monitoring systems represents a significant leap forward in the evolution of lithium battery technology, ensuring their continued relevance and reliability in various applications, including electric vehicles.

2.2. Parameter Identification for Deep-Cycle Batteries

Deep-cycle batteries are designed to meet the charging and discharging cycles that are required specifically in solar PV systems. These batteries have unique characteristics such as a high discharge rate, high power density, flat discharge curves and better performance in terms of low temperatures. For solar PV system configurations, maintenance-free lead–acid batteries are common since they are the most cost-effective. This subsection briefly outlines specific parameters linked to deep-cycle lead–acid batteries that are necessary for effectively monitoring and optimising the battery storage system, as outlined in [36].

Battery Current

This indicates the flow of electrons, and its rate per unit time is known as ampere. Batteries can store energy in the form of direct current (DC) to light up or power an inverter, which converts it into alternating current (AC). Usually, deep-cycle batteries are linked together in parallel to boost the total output current.

2.3. Battery Voltage

Voltage is known as the electrical pressure. For instance, the standard voltage for a car battery is 12 V as a result of the series connection of six (6) smaller lead–acid cells of 2 V each. Battery banks are primarily linked in series to generate DC voltages such as 12, 24, 36, or 48 V, making them suitable for various energy systems.

2.3.1. Rated Battery Capacity

This defines the energy capacity within a battery and is often quantified in amperehours (Ah) at a specific voltage. For example, a 100 ampere-hour rated battery can supply 10 amperes for 10 h, 1 ampere for 100 h and so on. Gauging this rated capacity repeatedly is challenging when these batteries are actively employed in PV systems.

2.3.2. Depth of Discharge

The depth of discharge signifies the proportion (%) of the battery's designated capacity that has been utilised. The construction of the battery is key for determining its capacity to withstand discharge. Battery construction can be categorised into shallow-cycle batteries (like most sealed, maintenance-free batteries) and deep-cycle batteries (which can endure repetitive charging and discharging to very high and low levels, respectively, without harming the cells).

2.3.3. Temperature Charts

Batteries are generally designed to be temperature sensitive and, as such, will not provide substantial power in a cold state compared to a warm condition. Manufacturers offer temperature correction charts to distributors and customers to address temperature-related impacts. For instance, a battery stored at 25 °C will exhibit 100% capacity when discharged at a rate equivalent to C/20, with the discharge rate typically calculated relative to the battery's rated capacity, C. It is advisable to maintain batteries at or near room temperature: around 25 °C.

2.3.4. Charging Cycle

Consideration of the optimal charging process for a battery involves several factors: firstly, charging with a consistent current until reaching a set voltage; secondly, maintaining a constant voltage while the charging current decreases; finally, reducing the charging voltage at an appropriate point to prevent excessive gassing and electrolyte loss. Achieving this ideal charging cycle is challenging in a PV system due to the continuous charging from available power. In standalone systems, the battery cycle typically spans 24 h: charging throughout the day and discharging during the night.

2.3.5. Battery Cycles

Predicting a battery's lifespan is highly challenging due to its reliance on various factors such as charging and discharging speeds, depth of discharges, cycle count and operational temperatures. The lifecycles of batteries in a PV configuration depend on the make and the conditions of operation.

2.3.6. Days of Autonomy

This indicates the duration during which the battery system can sustain the connected load without requiring a recharge from the photovoltaic array. This determination relies on factors such as the system itself, seasonal variations, the geographical location, the total connected load and the characteristics of the load within the system. The condition of the weather is also a major factor in determining the number of no-sun days.

2.3.7. State of Charge

The state-of-charge indicates the remaining capacity available in a battery at a specific moment.

$$SoC = 1 - DoD.$$
 (2)

3. Overview of SoC Estimation Methods

The realm of SoC estimation methods for batteries has undergone significant evolution, ranging from simplistic voltage-based approximations to complex model-based algorithms. While each approach has its merits, there persist substantial research gaps necessitating attention. Current methodologies often struggle with accurate SoC determination under varying operational conditions, such as high charge/discharge rates or extreme temperatures, demanding improved robustness and adaptability. Additionally, the impact of ageing on battery performance remains a critical area requiring comprehensive exploration. Existing methods often falter in accurately estimating the SoC as batteries degrade over

time, highlighting a need for innovative strategies to account for ageing effects to extend the accuracy and reliability of these estimations throughout a battery's lifecycle [69,70].

Furthermore, the integration of diverse estimation techniques or hybrid approaches emerges as a promising avenue for enhancing accuracy and reliability across different battery chemistries and applications. Combining the strengths of, for instance, modelbased algorithms with machine learning or data-driven approaches might yield more robust SoC estimations that are capable of handling nonlinearities and can mitigate the limitations observed in individual methods. However, this interdisciplinary approach also presents challenges, primarily in ensuring seamless integration and avoiding computational complexities that could hinder real-time application [71,72]. Addressing these gaps in research demands a concerted effort toward not only refining existing methodologies but also innovating new, comprehensive approaches capable of accurately estimating the SoC across diverse operating conditions and ageing scenarios.

As discussed in the previous section, knowing the battery's exact SoC is an important part of the BMS as it allows for the necessary assessment factor for battery management in photovoltaic systems. Therefore, techniques have been introduced to estimate the SoC of batteries. In this section, the different methods in the literature have been grouped and classified as shown in Figure 2. Additionally, the works of researchers who have utilised these methods for developing the SoC for batteries are highlighted in the subsequent subsections.



Figure 2. SoC estimation methods.

3.1. Direct-Measurement-Based Methods

These methods use the physical properties of batteries, like the impedance, terminal voltage and current, to measure the SoC [73]. They encompass techniques like the opencircuit voltage method, terminal voltage method, impedance method and impedance spectroscopy method.

3.1.1. Open-Circuit Voltage Method

As defined by [74], the open-circuit voltage (OCV) is the maximum voltage that a device can produce when it is illuminated. It is also described as the maximum voltage available that is drawn out from a solar cell and occurs at zero current while operating linearly to the SoC. In scenarios where the OCV is available, estimating the battery's SoC involves a direct measurement between the SoC and OCV. This relationship is examined to enhance battery management technologies and refine SoC estimations [75]. However, real-time measurement of OCV is challenging as it requires shutting off the power and allowing the battery to cool down for a specific duration [76]. Additionally, the OCV is

influenced by the cell ambient temperatures, and each cell in batteries possesses distinct chemical characteristics, leading to varied voltage profiles among cells [77]. Moreover, the OCV decreases in low temperatures compared to higher ones, impacting battery voltage measurements. This temperature dependency often introduces errors in SoC estimation during battery operation [78]. Therefore, achieving accurate SoC estimations requires allowing the battery sufficient rest to reach a cell equilibrium state before evaluating its OCV [78]. Employing this approach frequently during rest periods can be combined with other techniques to further validate the SoC indication [79].

Furthermore, when integrating various techniques, OCV measurement can fine-tune other approaches [80]. Ideally, when estimating a battery management system's SoC via the OCV approach, the battery should be in a floating state without a load for accurate assessment [81]. The OCV method incorporates the Randles battery model, encompassing parameters like internal resistance (R_s), double-layer capacitance (C_d) and polarisation resistance (R_{ct}), as depicted in Figure 3 [77].



Figure 3. Randles equivalent model of a standard battery, adapted with permission from Ref. [77].

The figure represents a battery terminal at the condition of no load. Based on the Randles equivalent circuit, the battery's terminal voltage $V_{terminal}$ can be presented as:

$$V_{terminal} = OCV(SoC) - i(R_s + R_{ct}(I - \exp^{-t/\tau}))$$
(3)

where the parameters can be calculated using τ .

$$R_s = \frac{V_1}{i}; R_{ct} = \frac{V_2}{i}; C_d = \frac{\tau}{R_{ct}}$$

where V_1 , V_2 and τ can be derived by the time-varying battery terminal voltage, as shown in Figure 4. The figure also exhibits the time constant within the RC parallel circuit ($\tau = R_{ct} = C_d$), denoting the duration for the battery terminal voltage to decline to 63.2% of its typical value [77]. As such, one can represent the charge of a battery as a function of its SoC. Consequently, the OCV method uses Equation (3) data along with OCV measurement to estimate the battery's SoC.

Through this method, Gismero et al. [82] successfully reset the SoC estimation process, preventing the accumulation of errors. Additionally, the SoH is determined by assessing the charge accumulated between two distinct SoC levels using the least squares method. Tian et al. [83] employed the OCV test to detect ageing in lithium–ion batteries, forming a foundation for SoC estimation. Their method estimated what they called electrode ageing parameters (EAPs), which are offline OCV-based ageing diagnosis results that include electrode capacities and initial SoCs. In order to assess EAPs without feature extraction, this method required daily charging profiles to be run through a convolutional neural network. Experiments conducted on eight cells demonstrated the method's efficacy at

swiftly detecting lithium–ion battery ageing. Dang et al. [84] introduced an OCV-based method for SoC estimation that utilised a dual-neural-network-fusion battery model. The model was trained using dynamic stress test data, establishing an offline relationship between the OCV and SoC. Results of the experiments showed that the first/second-order electrochemical model served as the basis for the SoC estimation in the dual-neural-network-fusion battery model.



Figure 4. Time-varying battery terminal voltage, adapted with permission from Ref. [77].

3.1.2. Terminal Voltage Estimation Method

This method relies on the decrease in terminal voltage caused by internal impedances during battery discharge. The battery's electromotive force (EMF) is directly correlated to the terminal voltage [85]. Moreover, a linear relationship between the EMF and the battery's SoC can be established [86]. During discharge, the battery's terminal voltage is contingent on the internal impedance voltage [68]. Thus, the battery's terminal voltage exhibits a linear correlation with the SoC. In this method, the SoC can be expressed mathematically as:

$$\frac{d_s}{d_t} = \frac{i}{36Q_a}; Q_a = Q_{a0} - Q_f \tag{4}$$

where S represents the cell's SoC, Q_a stands for the real-time capacity of the battery, Q_{a0} denotes the actual capacity of the battery, and Q_f refers to the capacity decline due to ageing effects [77]. Several researchers have employed this approach to determine the SoC in batteries. For instance, in the work by Ren et al. [87], the authors developed a battery management system aimed at detecting the SoC and enhancing battery performance. The terminal voltage of each cell is measured, and the balancing system is notified when the SoC falls below or exceeds the defined threshold, reducing the output voltage ripple. Sato et al. [85] proposed an equation based on the terminal voltage. Experiments on discharging of lead–acid batteries were conducted, and the results demonstrated acceptable performance in contrast to alternative approaches.

3.1.3. Electrochemical-Impedance-Based Method

The electrochemical-impedance-based method uses the battery's voltage and current to gauge its internal resistance. Voltage measurement involves tracking the current changes over a brief period [88]. The ratio between the voltage and current fluctuations yields the direct current resistance, which represents the battery's capacity in DC [88]. To capture the ohmic effect and lessen the influence of transfer reactions and acid diffusion, a brief

interval is usually required. Extended durations cause the estimated resistance value to be inaccurate [89]. Moreover, this method showcases adaptability and demonstrates higher accuracy in estimating the SoC during the latter stages of discharging [90]. However, due to its low value, obtaining precise internal resistance becomes challenging. The internal resistance varies significantly with a wide range of SoCs and is scarcely detectable, thereby complicating the estimation of the SoC using the DC internal resistance [91].

3.1.4. Electrochemical-Impedance-Spectroscopy-Based Method

The electrochemical impedance spectroscopy (EIS) method gauges battery impedances across a broad spectrum of AC frequencies under varying currents [92]. Widely utilised to explore electrochemical processes, it has been extensively studied for diverse battery systems, including for assessing the SoC [93]. Model impedance values are derived by fitting measured impedance values using the least squares methodology. EIS is non-destructive and measures parameters effectively. The battery management system (BMS) produces a distinct frequency sine wave current response when a sine wave voltage signal is applied to it. The system's electrochemical impedance spectrum is formed by the ratio between the excitation voltage and the response current [94].

Current battery impedances can be analysed and used to indirectly determine the SoC relative to known impedances across different SoC ranges [95]. Additionally, the EIS method yields more accurate state-of-health (SoH) estimations for lead-acid batteries compared to SoC estimations. For Li-ion batteries, EIS is initially used and proves practical for SoC estimation [96]. The ambient temperature notably affects battery impedance, particularly at low frequencies during electrochemical processes, thus requiring SoC and SoH estimation through EIS at higher frequencies.

In Xu et al.'s work [97], the authors introduced a way to estimate a lithium-ion battery SoC. They analysed impedance spectra from electrochemical impedance spectroscopy to represent the battery impedance using a constant-phase element. A fractional-order calculus technique was used to design this element in the impedance model. Subsequently, a Kalman filter was employed to evaluate the battery's SoC and showed satisfactory performance. Zhang et al. [98] suggested an EIS-based method for estimating the state-of-health (SoH) of batteries that takes temperature and SoC variations into account. Originally, to connect the SoH at various SoCs, an analogous circuit model employed the charge transfer resistance and solid electrolyte interface resistance. For Li-ion battery SoH estimation, they developed a probabilistic model integrating the temperature and SoC as input variables. Experiments revealed this approach's superiority over existing EIS-based techniques. In a different study [99], the authors measured and identified dependency levels by performing correlational analysis of equivalent circuit parts and impedance ranges of several commercial Li-ion batteries across different SoCs and internal temperatures. They used a curve-fitting approach to match the measured impedance spectra onto an equivalent circuit model. Impedance variables were revealed to be highly reliant on the SoC and internal temperature.

3.2. Bookkeeping-Based Methods

The bookkeeping estimation technique relies on the discharging current data from the battery as its input [91]. This method enables the incorporation of specific internal battery aspects like self-discharge, capacity degradation and discharge efficiency [100]. Techniques within this category encompass the coulomb counting method and the modified coulomb counting method, which are detailed in the subsequent subsections.

3.2.1. Coulomb Counting Method

The coulomb counting (CC) technique evaluates the discharge current of a battery and uses a time integration of this current to calculate the SoC [101]. It is also known as the ampere-hour method and operates on the premise of knowing a battery's maximum available capacity and evaluating its current. Through the ampere-hour integral method, it is possible to precisely calculate the SoC variation. Additionally, an exact SoC value can be obtained if the starting SoC is known. Mathematically, the SoC can be derived from Equation (5) [77]:

$$SoC_{t} = SoC_{t_{0}} + \frac{1}{C_{n}} \int_{t_{0}}^{t_{0}+t} I_{bat} \, d\rho,$$
(5)

where SoC_{t_0} represents the initial SoC of the battery, C_n denotes the battery's nominal capacity, and I_{bat} defines the charging and discharging current of the battery. The coulomb counting (CC) method's preciseness is impacted by various factors such as discharging current, temperature, battery lifespan and ageing, prompting the introduction of an improved CC method to refine the conventional approach [102]. This method proves effective for batteries without causing side effects during regular operation. However, employing this method for SoC estimation comes with three drawbacks [103]. Firstly, it necessitates consideration of the initial SoC. Secondly, inevitable measurement errors arise in the battery becomes necessary due to variations in operating conditions and battery usage levels [104].

In the study by Movassagh et al. [101], the issue of the SoC of rechargeable batteries was addressed. They outlined issues such as errors in current measurement, uncertainty regarding battery capacity, timing oscillator errors and approximations in current integration. They further recommended enhancement of the CC approach in order to solve these issues. Hence, due to these disadvantages of the CC method, researchers have envisaged to enhance the CC approach as discussed in the following subsection.

3.2.2. Modified Coulomb Counting Method

As stated in the previous subsection, the coulomb counting method calculates the SoC of a battery by integrating the current flowing into and out of the battery over time [105]. Although widely used due to its straightforward application, the precision of coulomb counting can be influenced by various factors, potentially leading to inaccuracies in SoC estimation. Key factors affecting the precision of coulomb counting include measurement errors and the initial SoC uncertainty [106]. Measurement errors primarily arise from inaccuracies in current sensing, where calibration, drift or offset errors in current sensors can significantly impact the results [107]. Even minor inaccuracies when integrated over time can lead to substantial errors in SoC estimation. Additionally, the sampling frequency can affect the method's precision, as infrequent sampling might fail to capture short-term fluctuations in current, especially in applications with variable load profiles [63]. The method also relies heavily on an accurate initial SoC estimate to begin calculations; any error in this initial value can propagate throughout subsequent calculations, leading to greater inaccuracies.

Moreover, the method does not inherently account for battery efficiency and ageing effects, which are crucial for accurate SoC estimations [106]. Charge efficiency varies and may not always be constant, as it depends on the battery's state and operational conditions. As batteries age, their capacity decreases, and internal resistance changes, further complicating precise SoC estimations. Coulomb counting typically does not include temperature effects directly, which significantly influences battery performance, including the charge efficiency and self-discharge rate [108].

To enhance the precision of coulomb counting, several improvements can be implemented. High-precision, low-drift current sensors can minimise measurement errors, and sensors with self-calibrating features may help maintain accuracy over time. Regular calibration of the system to account for sensor drift and other measurement system changes can also preserve the accuracy of SoC estimations. Additionally, integrating coulomb counting with other SoC estimation methods, such as voltage-based methods or impedance spectroscopy, can help offset some of its limitations [109]. For example, voltage measurements can periodically recalibrate the SoC estimation to correct any drifts arising from the coulomb counting method alone [110]. Incorporating temperature compensation algorithms can adjust SoC calculations based on temperature measurements, helping to account for temperature-dependent changes in battery efficiency and self-discharge rates [111]. Furthermore, employing machine learning algorithms can enhance the method's accuracy by predicting and adjusting for deviations caused by factors like ageing, temperature variations and load variability [112]. These models can utilise historical data to continually improve the precision of SoC estimations, making coulomb counting more reliable for modern battery management systems.

To refine the CC method, a modified version called the modified coulomb counting method was introduced. This approach utilises adjusted current values to enhance estimation accuracy, which are derived from the discharging current [113]. Moreover, there is a quadratic connection between the adjusted current and the battery's discharging current. Thus, through practical experimentation, the adjusted current is computed using Equation (6):

$$I_c(t) = K_2(t)2 + K_1I(t) + K_0,$$
(6)

where the constants K_0 , K_1 and K_2 are derived from empirical data obtained through practical experiments. In the modified coulomb counting method, the SoC is calculated using Equation (7).

$$SoC_t = SoC(t-1) + I_c(t)Q_n\delta_t$$
(7)

It has been confirmed in the literature that the modified coulomb counting method is more precise than the traditional coulomb counting approach [104]. Researchers have extensively employed the modified coulomb counting technique. In [114], the authors introduced a data-driven coulomb counting method for estimating the SoC of lithium-ion batteries. They adjusted the flawed initial SoC by transforming battery-voltage-based incremental capacity curves into SoC-based incremental capacity curves. Their approach was tested under conditions of rapid capacity degradation, demonstrating its ability to rectify inaccurate parameters and provide satisfactory SoC accuracy. Ko et al. [102] suggested a lithium-ion battery capacity estimation method centred around an enhanced coulomb counting methodology. However, their approach yields a wide error range in battery capacity estimation due to issues in SoC estimation, necessitating further methods to attain precise battery capacity values. Consequently, they introduced a Kalman filter and applied it to the capacity values estimated by their approach. Lithium-ion battery experiments demonstrated better performance compared to alternative approaches. In [115], the authors introduced a linear second-order extended state observer to solve the issues of the conventional coulomb counting method for estimating the SoC. Experimental results indicated that the suggested approach outperformed established conventional methods in performance evaluation.

3.3. Adaptive-Based Methods

A number of adaptive systems for SoC estimation have been developed as a result of developments in artificial intelligence (AI) and machine learning (ML). These methods encompass neural networks (NNs) and Kalman filters [116]. These systems possess self-designing capabilities, enabling them to intelligently adapt to changing environments or systems [117]. Considering batteries' susceptibility to diverse chemical influences and their nonlinear SoC behaviour, adaptive-based approaches offer solutions to SoC estimation challenges [77]. The different adaptive-based methods are discussed in the following subsections.

3.3.1. Machine Learning Modelling/Simulation

The application of machine learning algorithms for improving the accuracy and reliability of SoC estimation for batteries has seen significant advancements [118–120]. Supervised learning algorithms, for instance, involve training models on labelled datasets wherein the input features are paired with known outputs [121]. In the context of SoC

estimation, this typically means using historical battery data to predict the current SoC. A notable case study is the work by Zhang et al., which utilised support vector machines (SVMs) to estimate the SoC of lithium–ion batteries [122]. Their approach trained the SVM model using features such as voltage, current and temperature data. The results indicated that the SVM model demonstrated high accuracy and robustness, outperforming traditional methods like coulomb counting and model-based estimations. One of the main advantages highlighted was the SVM's ability to provide better generalisation to unseen data and to effectively handle nonlinear relationships [122].

Reinforcement learning (RL) presents another promising avenue for SoC estimation. RL involves an agent learning to make decisions by interacting with the environment to maximise cumulative rewards [123,124]. For SoC estimation, RL algorithms can optimise the estimation process by continuously learning and adapting to battery behaviour. Liu et al., demonstrated this through the application of deep Q-learning, a type of RL, to estimate the SoC dynamically [65]. Their method involved training the RL agent to predict the SoC by receiving rewards for accurate predictions and penalties for errors, using data such as charge/discharge cycles and temperature. The RL-based model showed significant improvements in adaptability and accuracy compared to static estimation methods. The primary advantage of RL models lies in their ability to adapt to changing conditions in real-time, providing more accurate and reliable SoC estimates under varying operational conditions [65].

Deep learning, particularly using neural networks, has also shown great promise in SoC estimation due to its ability to model complex and nonlinear relationships in data [125]. Chen et al. implemented long short-term memory (LSTM) networks to estimate the SoC of lithium–ion batteries [126]. The LSTM model was trained on sequential data such as voltage, current and temperature and captured temporal dependencies effectively. The results were impressive, with the LSTM model achieving superior performance in terms of accuracy and reliability, significantly reducing the root mean square error (RMSE) compared to traditional methods. Deep learning models, particularly LSTM networks, excel at handling time-series data and can capture long-term dependencies, leading to more precise SoC estimations [126].

Traditional methods of SoC estimation, such as coulomb counting and model-based methods, have several limitations. Coulomb counting tracks the charge/discharge current over time but suffers from cumulative error, while model-based methods use equivalent circuit models or electrochemical models but require precise parameter identification and are sensitive to model inaccuracies [105,127]. In contrast, machine learning approaches offer several advantages. They improve accuracy by learning complex patterns and relationships in data that traditional methods might miss [128]. They also provide adaptability, as machine learning models can continuously learn and adapt to new data, improving estimation under varying conditions [129]. Additionally, machine learning models reduce complexity as, once trained, they can provide real-time SoC estimation without the need for complex calculations or parameter identification [130].

In conclusion, machine learning algorithms, including supervised learning, reinforcement learning and deep learning, have significantly enhanced the accuracy and reliability of SoC estimation. Their ability to model nonlinear relationships, adapt to new data and provide real-time estimations offers substantial advantages over traditional methods. Recent research and case studies underscore the potential of these advanced algorithms for optimising battery management systems and enhancing the performance and lifespan of battery-powered devices.

3.3.2. Neural Network

A neural network (NN) applies a mathematical algorithmic model to handle tasks involving simultaneous processes [131]. It excels at processing data and resolving relationships among various initially complex factors [132]. Among NNs, a backpropagation neural network, a type designed to solve nonlinear problems, offers a simpler structure

compared to standard neural network methods [133]. A three-layered backpropagation neural network (input, hidden and output) is capable of estimating the SoC of batteries. Various factors, including battery voltage, current, resistance and ambient temperature are incorporated into the input layer. The output layer provides an estimated range value for the SoC, and the number of hidden layers is dependent on the precision of the system [134]. The primary objective of this SoC estimation approach is to minimise the margin of error. However, this error hinges on the volume of training data utilised during experiments [77]. Training data used for SoC estimation is derived from charging and discharging battery experiments. These errors escalate when the backpropagation neural network lacks sufficient training data encompassing SoC values [135]. Hence, achieving an accurate SoC value necessitates employing a large dataset from various batteries, as battery discharge characteristics can vary based on electrolyte volumes.

Figure 5 gives a general neural network structure for SoC estimation, and Table 4 gives a comparison of varied neural network structures used for SoC estimation. The application of neural networks in SoC estimation for batteries involves configuring input, hidden and output layers to enhance the accuracy and reliability [136]. The input layer receives features like voltage, current and temperature, while hidden layers with multiple neurons capture nonlinear relationships using activation functions such as ReLU, sigmoid and tanh [137]. The output layer, typically with one neuron, provides the continuous SoC value.



Figure 5. Neural network structure, adapted with permission from [138].

In [139], the authors proposed a comprehensive equivalent circuit model integrating neural networks to address battery model issues in real-time. Additionally, a radial basis function neural network was employed for estimating battery SoC. An experimental evaluation pitted this approach against an extended Kalman filter, where the former showcased faster convergence speed and higher precision. Yang et al. [140] developed a recurrent neural network with gated recurrent units to estimate battery SoC centred on the assessed current, voltage and temperature. Their approach demonstrated improved estimation accuracy compared to conventional neural network methods. Huang et al. [141] proposed a convolutional gated recurrent unit (CNN-GRU) network for estimating lithiumion battery SoC. The training for this method was done using data from battery discharge processes. Experimental results revealed that their proposed technique outperformed other deep learning methods for lithium—ion battery SoC estimation.

	-					
Neural Network Structure	Number of Hidden Layers	Type of Machine Learning/AI	Inputs	Training Algorithm	Performance Metrics	Reference
Feedforward Neural Network (FNN)	2–4	Supervised Learning	Voltage, Current, Temperature	Backpropagation	Root Mean Square Error (RMSE), Mean Absolute Error (MAE), Coefficient of Determination (R^2)	[142,143]
Recurrent Neural Network (RNN)	3–5	Supervised Learning	Voltage, Current, SoC History	Adam Optimizer	Root Mean Square Error (RMSE), Accuracy	[125,144]
Convolutional Neural Network (CNN)	4–6	Deep Learning	Battery Images (Thermal), Voltage	Stochastic Gradient Descent	Accuracy, F1 Score	[145]
Hybrid Neural Network (CNN-LSTM)	3–5	Deep Learning	Voltage, Current, Temperature, SoC History	Adam Optimizer	Root Mean Square Error (RMSE), Mean Absolute Error (MAE), Coefficient of Determination (R^2)	[146]
Autoencoder Neural Network	3–4	Unsupervised Learning	Voltage, Current, Temperature	Adam Optimizer	Root Mean Square Error (RMSE), Mean Absolute Error (MAE)	[147]
Radial Basis Function Neural Network (RBFNN)	1–3	Supervised Learning	Voltage, Current, State of Charge	K-means Clustering + Gradient Descent	Mean Square Error (MSE), Root Mean Square Error (RMSE)	[148]
Extreme Learning Machine (ELM)	2–4	Supervised Learning	Voltage, Current, Temperature	Analytical Solution	Mean Absolute Error (MAE), Coefficient of Determination (R^2)	[149]

Table 4. Comparison of neural network structures for SoC estimation.

3.3.3. Kalman Filter

The Kalman filter used for SoC estimation relies on an algorithm that utilises precisely measured state variables to account for time-varying noise [150]. This filter estimates the SoC by creating a model of the battery system and other necessary parameters required for SoC measurement. Essentially, it functions as an algorithm that is employed to estimate the internal states of dynamic systems [151]. In the context of battery management systems (BMSs), the battery represents the dynamic system, and the SoC represents the internal state [152]. The Kalman filter intelligently offers estimated values within dynamic states, making it suitable for applications involving dynamic models [153]. However, owing to batteries' nonlinear traits, an extended Kalman filter (EKF) approach becomes necessary. This method necessitates a battery model that is capable of representing the dynamic state [79]. Despite its ability to estimate battery SoC during operational phases, the EKF method encounters a drawback wherein the SoC estimation duration increases as the number of state variables rises [154]. Additionally, since the EKF method relies on linearising the approximation of nonlinear functions using first- or second-order terms from complex formulas, it often fails to deliver precise performance [77]. To solve this, an unscented KF has been presented and was shown to have higher accuracy compared to the EKF [155]. Regardless of the KF method adopted, it is advised that the statistical information of battery noise is accurate to have an effective SoC estimation.

In the work presented in [150], the authors introduced a fractional-order adaptive extended Kalman filter (FO-AEKF) designed for SoC estimation. They proposed a fractional-order model to characterise battery behaviour, asserting its superiority over integral-order models due to its decentralised and diffuse characteristics for battery internal parameters. Using a genetic algorithm, they identified the parameters of this fractional-order model, ensuring adequate parameter identification. The authors claim that their method swiftly detects unknown and time-invariant factors, thereby enhancing SoC estimation accuracy. A different strategy presented in [156] conveys an adaptive square-root sigma-point Kalman filter (ASR-SPKF)-based SoC estimation technique built around a reduced-order electrochemical model. Their approach refines the estimation of anode and cathode SoCs by integrating constraints based on the principle of charge conservation. Evaluations across three filters considering estimation accuracy, error margins and computational time revealed satisfactory performance of their proposed approach.

Table 5 offers additional information for an in-depth analysis of the benefits and drawbacks of the various SoC techniques covered in this study.

Method	Applicability	Advantages	Disadvantages
Open-circuit voltage [83]	Lithium–ion battery; lead–acid	Easy to use; accessible online	Long rest time is required; sensitive to temperature.
Terminal voltage [157]	Most energy storage	Easy to use; accessible online	Error in estimation due to drop in terminal voltage at the end of discharge.
Electro-impedance spectroscopy [158]	Most energy storage	Easy to use; accessible online; can estimate most of the battery parameters	Sensitive to temperature; high frequency is required.
Coulomb counting [159]	Most energy storage	Easy to use; low computational cost	Current leakage during charging, thereby affecting the estimation of SoC.
Modified coulomb counting [114]	Most energy storage	Easy to use; low computational cost	Current leakage during charging, thereby affecting the estimation of SoC.
Neural network [160]	Most energy storage	Adapts to time-varying characteristics; fast computational cost for online phase	Requires a lot of training data for accuracy; network convergence is slow; difficult to achieve global optimum.
Modified Kalman filter [153]	Most energy storage	Accessible online; dynamic	Battery model is required; initial parameter issues.

Table 5. Summary of the advantages and disadvantages of various SoC methods discussed.

3.4. Factors Affecting the Battery SoC

There exist factors that disrupt the lifespan and performance of photovoltaic batteries [66]. The effectiveness of batteries can be derived from the availability of energy and the lost energy or battery capacity. Batteries are considered to be less effective when the effective materials are changed into their inactive phases through the process of charging and its chemical reactions [161]. Knowing the factors that can affect the SoC of batteries is important for their lifespan. Hence, this section highlights different factors that can affect the SoC of batteries. These include battery age, temperature, charge current, internal resistance, battery temperature, charge/discharge depth and charge/discharge rate [73].

3.4.1. Battery Age

The older the battery is, the less effective it becomes in terms of performance [162]. When batteries become old, they experience irreversible capacity loss due to liquid electrolytes drying up and a lithium–ion layer building up on the electrodes [163]. Also, the ageing of the battery results in diminished capacity, increased resistance and elevated self-discharge. All these reduce the battery SoC, thereby affecting the performance of batteries [164].

3.4.2. Temperature

Batteries' lifespans and performance depend on the process of charging and discharging, where temperature plays an important role [165]. Overheating of the batteries results in thermal runway, where the temperature can reach more than 400 degrees Celsius. A single thermal runway leads to a reaction with other cells, causing life or property loss [166]. To enhance battery longevity and ensure safety measures, batteries are equipped with a battery management system overseeing the charging, discharging and overall temperature regulation process [167]. Moreover, for added protection during charging and discharging, a thermal management system can be employed [168]. This thermal control mechanism serves to regulate the temperature across the entire battery system and encompasses cooling mechanisms and heating sources to warm batteries in cold environments [169].

3.4.3. Charge Current

The process of charging batteries causes changes within its internal chemistry [151]. Most importantly, charging batteries at a high current worsens these effects [151]. The ideal battery charging process should be in a manner in which the current is managed at a level that increases the battery's efficiency and lifespan [6]. When batteries are constantly charged at a high current, electrolytes and atoms protecting the anode become thickened, which later affects ions' access to the anode. Similarly, on the cathode, there is a gathering of lithium ions that causes electrolyte oxidation, which can cause thermal runaway [170]. Furthermore, charging the battery at a low current can increase its lifespan. However, it can reduce the battery's time performance [171].

3.4.4. Internal Resistance

Variations in factors like age, current and size can impact a battery's internal resistance [172]. A decrease in the internal resistance enhances the battery's efficiency [173]. Photovoltaic batteries usually have one of the lowest internal resistances, and the electrolyte interface layer improves the internal resistance [75]. The layer assists with the battery process since it manages the system and improves its lifespan. However, this can result in more internal resistance over time [174].

3.4.5. Charge/Discharge Depth

When batteries are unable to charge beyond a certain percentage of their initial capacity, this signifies that they have reached the conclusion of their practical lifespan [137]. The depth of discharge plays a role in determining the quantity of charge cycles a battery can provide throughout its useful existence [175]. Additionally, the depth of discharge serves as an indicator of the battery's health, offering an alternative to the SoC for assessment

purposes [176]. To manage battery life, researchers have come up with applications that can be embedded within battery management systems to decrease the problem of allowing batteries to drain and charge above certain particular levels [177].

3.4.6. Charge/Discharge Rate

Experts from both industry and academia recommend maintaining a constant current and voltage during the battery charging process [178]. This involves sustaining a steady current throughout the charging phase, ensuring the system delivers consistent current until the battery reaches its maximum charging voltage, after which the current is reduced to prevent overcharging the cells [137]. The idea is to maintain a manageable temperature and to avoid lithium coating of the electrodes.

3.5. Comparative Analysis of SoC Estimation Approaches

Accuracy, complexity, real-time implementation and trade-offs are the main criteria used when comparing different SoC estimation methods due to their critical roles in battery management. Accuracy ensures precise SoC readings: vital for preventing overcharging or deep discharging, which can degrade battery life and performance [179]. Complexity reflects the computational resources required, impacting the feasibility of deploying the method in various systems [180]. Real-time implementation assesses how well the method can operate within the constraints of real-time systems, ensuring timely and efficient battery management [114]. Trade-offs measure the method's ability to handle varying battery conditions and environmental changes, ensuring reliable performance under different scenarios [181]. These criteria collectively ensure that the chosen SoC estimation method provides reliable, efficient and effective battery management across diverse applications.

SoC estimation is vital for effective battery management in various applications. Different methods have been developed, each with its unique advantages and trade-offs in terms of complexity and accuracy. Coulomb counting is a basic method that measures the battery current over time to calculate the SoC [114]. It is straightforward to implement in real-time systems but is prone to cumulative errors due to sensor inaccuracies and integration drift, which can lead to significant deviations over extended periods [114,182]. Open-circuit voltage (OCV) methods rely on the relationship between the battery's opencircuit voltage and its SoC, providing accurate results when the battery is at rest [183]. However, this method requires the battery to be in an equilibrium state for accurate measurements, limiting its real-time applicability in dynamic environments [184,185].

More advanced techniques such as the Kalman filter (KF) and extended Kalman filter (EKF) offer higher accuracy by modelling the battery's dynamic behaviour and filtering out noise [186]. The KF strikes a balance between accuracy and complexity, making it suitable for real-time applications with moderate computational demands [187]. The EKF handles nonlinearities better than the KF, enhancing accuracy further but at the cost of increased computational requirements [182,188]. Particle filters provide very high accuracy by representing the probability distribution of the SoC as a set of particles. This method, while highly accurate, is computationally intensive and challenging to implement in real-time systems without substantial processing power [189]. Neural networks leverage machine learning to adapt to various battery conditions, offering high accuracy. However, they require extensive training data and significant computational resources, posing challenges for real-time implementation [108,189]. Each method presents distinct trade-offs between complexity and accuracy, necessitating careful consideration based on the specific requirements and constraints of the application [190,191]. Table 6 gives a summary of the comparison of the SoC estimation approaches.

Method	Description	Complexity	Accuracy	Implementation	Trade-Offs	References
Coulomb Counting	Measures current over time to estimate SoC.	Low	Medium	Simple	Simple implementation but susceptible to cumulative errors.	[114,182]
Open-Circuit Voltage (OCV)	Uses battery's open-circuit voltage to determine SoC.	Low	Medium	Simple	Requires resting periods for accurate measurement.	[184,185]
Kalman Filter	Models dynamic behaviour and filters noise.	Medium	High	Moderate	Balances accuracy and complexity well.	[187,190]
Extended Kalman Filter	Extends KF to handle nonlinearities in the battery model.	Medium	High	Moderate	Improves the Kalman Filter but with increased complexity.	[182,188]
Particle Filter	Uses a set of particles to represent the probability distribution of the SoC.	High	Very high	Complex	Highly accurate but computationally intensive.	[189,191]
Neural Network	Leverages machine learning to adapt to various conditions.	High	Very high	Complex	High accuracy and adaptability but requires extensive training data.	[108,189]

Table 6. (Comparative	analysis o	of SoC estin	nation app	roaches.
------------	-------------	------------	--------------	------------	----------

The incorporation of specific hybrid models for SoC estimation can enhance accuracy and reliability under variable operating conditions. Preliminary data and/or theoretical analyses that demonstrate the potential benefits of some of these hybrid approaches are discussed in the subsections below and are summarised in Table 7.

3.5.1. Kalman Filter (KF) + Artificial Neural Network (ANN) Hybrid Model

A Kalman filter (KF) + artificial neural network (ANN) hybrid model leverages the robustness of the Kalman filter and the adaptive learning capabilities of an artificial neural network to enhance the accuracy of SoC estimation. The Kalman filter is instrumental in providing real-time SoC estimation by effectively filtering out noise and uncertainties from measurement data, making it highly reliable in dynamic environments [160]. On the other hand, the ANN is trained on historical battery data, allowing it to learn complex patterns and predict the SoC under various operating conditions, particularly during rapid load changes or temperature variations [192]. The KF provides precision in real-time applications by accounting for noise and system uncertainties, which is crucial for maintaining reliability. However, its performance can degrade under nonlinear conditions or when rapid changes occur. The ANN addresses these limitations by compensating for model inaccuracies and nonlinearities, thereby enhancing the overall accuracy and robustness. Preliminary simulations using this hybrid approach have shown a reduction in SoC estimation error by approximately 15% compared to using the Kalman filter alone [160]. This improvement is most significant in environments with high variability, such as during rapid charging or discharging cycles, or when the battery operates under fluctuating temperatures, where the ANN's ability to predict the SoC based on learned patterns plays a crucial role in maintaining estimation accuracy.

3.5.2. Extended Kalman Filter (EKF) + Particle Filter (PF) Hybrid Model

An extended Kalman filter (EKF) + particle filter (PF) hybrid model is designed to improve SoC estimation accuracy in environments characterised by nonlinearities and timevarying conditions. The EKF, an extension of the standard Kalman filter, is used to linearise around an estimate of the current's mean and covariance, making it efficient for real-time estimation in near-linear systems [193]. However, its performance can be compromised in highly nonlinear regions. To address this, the particle filter (PF) is integrated into the model. The PF is a sequential Monte Carlo method that uses a set of particles to represent the posterior distribution of the system states, providing a probabilistic estimate of the SoC that is more accurate in nonlinear conditions [193]. This combination allows the EKF to efficiently estimate the SoC in near-linear regions, while the PF enhances accuracy in nonlinear areas by offering a flexible and robust approach to handle nonlinearities and non-Gaussian noise. Theoretical analyses and simulations suggest that this hybrid model can improve SoC estimation accuracy by up to 20%, particularly in scenarios involving rapid changes in load and temperature [194]. This improvement is particularly valuable in applications such as electric vehicles and grid storage systems, where operating conditions can vary significantly and maintaining high accuracy of SoC estimation is critical for system reliability.

3.5.3. Adaptive Observer + Fuzzy Logic Hybrid Model

The adaptive observer + fuzzy logic hybrid model is designed to enhance the accuracy and reliability of SoC estimation, particularly under conditions of uncertainty and variability. The adaptive observer is responsible for real-time SoC estimation and continuously adjusts its parameters based on feedback from the system to track the true SoC more accurately [195]. This adaptability is crucial in scenarios where the battery is subjected to fluctuating loads and temperatures, which can significantly affect its performance. Fuzzy logic is incorporated to manage the uncertainties and imprecise data that are often encountered in real-world applications [196]. Fuzzy logic allows the model to handle approximate reasoning, providing a more nuanced estimation of SoC when exact data are not available or when battery behaviour is unpredictable. This combination ensures that SoC estimation remains reliable even when the battery operates under harsh or unpredictable conditions, making the model particularly suitable for applications where precision is critical, such as aerospace or military systems. Preliminary simulations indicate that this hybrid approach can reduce the maximum estimation error by 10–15%, especially in environments with fluctuating temperatures and varying load demands [197]. The adaptive observer ensures that the model remains responsive to real-time changes, while fuzzy logic compensates for the uncertainties that could otherwise lead to inaccurate SoC estimates.

3.5.4. Model Predictive Control (MPC) + Machine Learning (ML) Hybrid Model

A model predictive control (MPC) + machine learning (ML) hybrid model integrates the strengths of model predictive control for real-time optimisation and machine learning for predictive capabilities based on historical data. MPC is used to optimise the SoC estimation process by considering future states and adjusting control inputs accordingly, ensuring that the system remains on an optimal path even as operating conditions change [198]. This is particularly important in real-time applications where the system dynamics are complex and require constant adjustment to maintain accuracy. Machine learning algorithms that have been trained on large datasets provide predictions based on long-term trends and patterns, which can be used to correct and refine the MPC's estimates [199]. The integration of ML allows the hybrid model to learn from past data, improving its ability to predict the SoC under varying conditions that MPC alone might not fully capture. Initial theoretical analyses and simulations suggest that this hybrid model can reduce SoC estimation errors by 15–20% compared to using MPC alone [200]. This is especially beneficial in applications involving complex and dynamic operating environments, such as in grid storage systems or electric vehicles, where both short-term and long-term accuracy are critical for maintaining system efficiency and reliability.

Hybrid Model	Model Overview	Theoretical Benefits	Preliminary Data	References
Kalman Filter (KF) + Artificial Neural Network (ANN)	Combines KF for real-time SoC estimation with ANN for predicting SoC under varying conditions.	KF handles noise and uncertainties, while ANN predicts SoC during rapid load changes and temperature fluctuations.	Reduces estimation error by approximately 15% under dynamic conditions.	[160,192,201]
Extended Kalman Filter (EKF) + Particle Filter (PF)	Integrates EKF for linearised state estimation with PF for probabilistic estimates in nonlinear regions.	EKF estimates in near-linear regions; PF improves accuracy in highly nonlinear regions.	Improves SoC estimation accuracy by up to 20% during rapid load and temperature changes.	[193,194]
Adaptive Observer + Fuzzy Logic	Uses an adaptive observer to estimate SoC and fuzzy logic to handle uncertainties and imprecise data.	Adaptive observer adjusts in real-time, while fuzzy logic adds robustness in uncertain conditions.	Reduces maximum estimation error by 10–15%, especially under fluctuating temperature and load.	[195–197]
Model Predictive Control (MPC) + Machine Learning (ML)	Integrates MPC for dynamic optimisation with ML to predict SoC based on historical data and patterns.	MPC optimises in real-time; ML corrects based on long-term trends, improving accuracy.	Reduces SoC estimation errors by 15–20% in complex, dynamic environments.	[198–200]

Table 7. Proposed hybrid models for SoC estimation.

3.6. Deeper Technical Analysis of the Most Promising Techniques

SoC estimation for lithium–ion batteries is crucial for ensuring their efficient and safe operation. Among the most promising techniques are Kalman filters, machine learning algorithms and hybrid methods, each offering unique advantages in terms of accuracy and implementation. Kalman filters, including the extended Kalman filter (EKF) and unscented Kalman filter (UKF), are widely used due to their robust performance in handling system noise and model uncertainties [179]. EKFs extend the basic Kalman filter to accommodate nonlinearities in the battery model, enhancing accuracy while maintaining computational efficiency [180]. Successful implementations include their use in electric vehicles (EVs) to provide real-time SoC estimation under varying operational conditions, significantly improving battery management systems (BMSs) [179,202,203].

Machine learning approaches, such as neural networks and support vector machines, leverage historical data to predict the SoC. These data-driven models can adapt to diverse battery conditions without requiring detailed physical models [114]. For instance, neural networks have been successfully employed in grid storage applications, demonstrating high accuracy in SoC prediction by learning from extensive datasets of charging and discharging cycles [114]. This adaptability makes machine learning approaches highly effective in dynamic environments where battery behaviour can vary significantly.

Hybrid methods combine the strengths of model-based and data-driven approaches. A notable example is the integration of the Kalman filter with machine learning techniques, which enhances the robustness and accuracy of SoC estimation [181]. These hybrid systems have been effectively implemented in renewable energy systems, where they manage the intermittent nature of power generation and storage, ensuring reliable energy supply and optimal battery usage [181,204]. The combination of the real-time correction capabilities of Kalman filters and the predictive power of machine learning provides a comprehensive solution for SoC estimation [83,205].

Overall, these advanced SoC estimation methods not only improve the precision of battery management systems but also extend the lifespan and efficiency of lithium– ion batteries, which is crucial for their widespread application in the automotive and energy sectors. Accurate SoC estimation is essential for preventing overcharging and deep discharging, which can significantly degrade battery performance and longevity [114,181]. As battery technologies continue to evolve, the integration of these advanced estimation techniques will play a vital role in ensuring their optimal performance and reliability.

4. State-of-the-Art Modelling and Simulation Tools

State-of-the-art modelling and simulation tools for battery systems have made significant strides, offering invaluable insights into the complex electrochemical processes within batteries. These tools range from physics-based models that account for the intricate chemical and physical phenomena to empirical and data-driven models that leverage large datasets for predictive analysis. While these tools have facilitated understanding and prediction to a considerable extent, there are notable research gaps that need attention. One significant gap lies in the integration of multiscale modelling approaches, which aim to bridge the gap between the atomic/molecular level processes and the macroscopic behaviour of batteries. Current models often focus on either microscale or macroscale aspects, lacking seamless integration between these scales, hindering a comprehensive understanding of battery behaviour under various operating conditions [206,207].

Additionally, the robustness and adaptability of these models across diverse battery chemistries and configurations remains a critical challenge. While certain models excel at predicting behaviour for specific battery types, their applicability and accuracy might diminish when applied to different chemistries or novel electrode designs. Enhancing the versatility of modelling tools to encompass various battery chemistries, electrode materials and architectures would greatly benefit the design and optimisation of next-generation batteries. Furthermore, the validation and calibration of these models with real-world data, especially under dynamic and non-standard operating conditions, pose ongoing challenges that require meticulous attention to ensure accurate predictions and reliable simulations [208,209]. Closing these gaps necessitates concerted efforts in developing versatile, multiscale models while rigorously validating and calibrating them across diverse battery systems and operating conditions.

Efforts in research are directed toward model-based analysis of stationary storage systems tailored for specific storage technologies and applications [210–212]. However, most of these endeavours are typically executed, tested and evaluated using simulation platforms due to the absence of available test environments and ongoing real-time projects for development and implementation. Moreover, there is a range of individually developed software aimed at techno-economic modelling of grid-based stationary storage systems, as different approaches have distinct objectives. Hence, it is crucial not to generalise all potential battery storage systems under a single commercial software platform. As a result, this section explores notable cutting-edge modelling and simulation tools applicable for modelling, testing and analysing battery storage systems. It is worth mentioning that this is by no means an all-encompassing and comprehensive compilation of all pertinent tools necessary in the context of this subject matter. Furthermore, alongside the forthcoming modelling tools, it is noteworthy that open modelling approaches and an extensive array of accessible open-source tools have surfaced within the energy sector. These resources are accessible without restriction for both industrial and academic research, fostering collaborative model toolkit development within a broader community [210–212].

4.1. MATLAB/Simulink

MATLAB is a programming and numerical computation platform used by researchers and scientists based on 4th-generation programming languages (C, C++, and Java) for analysing data, developing algorithms and creating user interfaces [213]. This tool is commonly used to solve problems related to energy management, scheduling issues for home appliances and operations, analyses, multi-objective energy management and microgrid modelling problems. Simulink represents a graphical data flow of the programming language. This package is based on a customised set of block libraries that can easily be integrated with the rest of the MATLAB environment. It is applicable for use in simulating, modelling, digital signal processing and control theories applied to multi-objective

27 of 44

dynamic systems [214]. In micro-girds, this package is applicable for solving problems related to modelling and simulation, optimal automation levels and long-term economical optimisation planning [215–217].

MATLAB/Simulink is renowned for its robust scalability and user-friendliness, making it a preferred tool for both small-scale and large-scale projects, including battery systems. It accommodates complex simulations involving extensive datasets and multiple variables, making it highly adaptable for various applications, such as the modelling and control of battery systems in electric vehicles and grid storage solutions [218]. The software offers an intuitive graphical interface that allows users to build models using block diagrams and provides extensive documentation, tutorials and user communities for support. This ease of use is particularly beneficial for designing battery management systems (BMSs), as it allows engineers to simulate different charging and discharging scenarios, thermal behaviours and state-of-health estimations [219]. Its drag-and-drop functionality enhances usability, and seamless integration with other MathWorks products facilitates its application in diverse engineering and scientific projects, such as integrating battery systems with renewable energy sources or electric grid applications [218]. MATLAB/Simulink supports various file formats and protocols, ensuring easy integration with existing systems, which is essential for comprehensive engineering analyses and developing sophisticated battery system models that can be tested and validated before deployment [218]. MAT-LAB/Simulink requires substantial computational resources for large-scale simulations due to its capability to handle complex models and extensive datasets, often necessitating high-performance hardware for optimal performance [218]. For typical applications, it may require a multi-core processor, 16–32 GB of RAM and a dedicated graphics card to efficiently handle simulations [220]. It integrates seamlessly with a wide array of existing systems and supports various file formats and protocols, facilitating its use across diverse technological environments.

4.2. StorageVET

The financial viability of utility-scale storage is evaluated using the Storage Value Estimation Tool (StorageVET), which was developed by the Electric Power Research Institute in the United States [221]. Users can easily compute estimates for battery energy storage systems (BESSs) using pre-established technical and financial input parameters by utilising this web-based tool. StorageVET allows for fair comparisons in terms of profitability for different storage projects and grid services. It cuts across various technologies and sizes and also caters to varied locations. Based on battery specification modelling, StorageVET is strictly based on formalised parameters such as SoC constraints, efficiency of the battery and cycle and calendar lifetime estimations and, as such, does not support lithium–ion battery model specifications [222].

StorageVET is designed to evaluate the economic value of energy storage systems and is capable of handling both small- and large-scale battery storage projects effectively [222]. Its web-based interface simplifies data input and analysis, making the tool user-friendly, especially for stakeholders looking to evaluate the financial viability of different battery storage configurations and operational strategies. While the interface is straightforward, new users might require some time to become familiar with its economic evaluation functionalities. StorageVET integrates well with other energy management systems, facilitating the analysis of energy storage in various contexts, such as evaluating the cost-benefit of installing battery systems in commercial buildings, utility grids or residential setups [222]. This integration capability ensures that users can assess the potential savings, return on investment and environmental impact of different battery technologies and deployment strategies, making it adaptable for different scales of implementation. StorageVET is designed to be computationally efficient, but the complexity of the simulations can increase computational demands, especially for large-scale energy storage evaluations [223]. It typically runs effectively on systems with at least a quad-core processor, 8-16 GB of RAM and stable internet access for its web-based operations. Its web-based nature allows for

easy integration with existing energy management systems and data sources, although initial setup and data entry might require some effort [224].

4.3. DER-VET

An open-source program called the Distributed Energy Resource Value Estimation Tool (DER-VET) is intended to help determine, comprehend and maximise the value of distributed energy resources (DERs) in accordance with their technical advantages and limitations. With DER-VET, the EPRI's StorageVET tool is expanded to include site-specific evaluations of energy storage as well as other distributed energy resources (DERs) like solar, wind, demand response, EV charging, internal combustion engines, combined heat and power systems as well as microgrids. It uses load data, location-based parameters and financial benefits that can be gathered from specific use cases to determine the optimal size, duration and other parameters for maximising benefits [225].

DER-VET is scalable and can be used for analysing the value of distributed energy resources (DERs) across different project sizes, from small community installations to large utility-scale projects [226]. It is particularly useful for assessing the role of battery systems in enhancing grid stability, providing backup power and supporting renewable energy integration. Designed with user-friendliness in mind, DER-VET offers a clear interface and comprehensive guides to help users conduct detailed evaluations of DERs, including various types of battery storage systems. The web-based interface is intuitive, allowing users to easily navigate through the evaluation process and understand how battery systems can optimise energy use, reduce costs and improve reliability [226]. DER-VET supports integration with various data sources and energy management systems, making it versatile for different applications and adaptable to varying project scales. This makes it an excellent tool for planners and engineers looking to integrate battery storage into broader energy systems and to ensure efficient operation and to maximize benefits [226]. DER-VET has moderate computational demands, is suitable for evaluating distributed energy resources across various scales and can be run on standard computing hardware [227]. A typical setup might include a multi-core processor, 8–16 GB of RAM and a standard graphics card. It is highly compatible with existing energy management systems and integrates well with multiple data sources, enhancing its utility in diverse applications [228].

4.4. HOMER

The Hybrid Optimisation Model for Multiple Energy Resources (HOMER) was initially created by the National Renewable Energy Laboratories (NREL) in the United States. HOMER is a modelling tool that supports the technical and economic optimisation of microgrid systems and distributed energy resources. This software enables end-users in the ever-growing microgrid and distributed energy market to efficiently maximise costs and revenues in line with the hybrid microgrid value chain. The focus is not on the battery itself but on the entire grid system. HOMER comprises three powerful tools in a single software suite, including HOMER Pro, HOMER Grid and HOMER Front [229,230].

HOMER is highly scalable and is capable of modelling and optimizing both small off-grid systems and large grid-tied installations, including those that incorporate battery storage solutions [229]. The software is considered user-friendly, with a graphical user interface that simplifies the process of system design and optimisation. This feature is particularly advantageous for designing and evaluating hybrid energy systems wherein battery storage plays a crucial role in balancing supply and demand, storing excess renewable energy and providing power during outages. Extensive tutorials and user support enhance its accessibility for users of all experience levels, making it easier for users to simulate different scenarios and determine the optimal sizes and configurations of battery systems for specific applications [229]. The graphical interface supports drag-and-drop functionality for system components, making it easy to use and allowing for quick adjustments to the system design as new data and technologies become available. HOMER integrates well with other tools and data sources, allowing users to import and export data easily,

which is essential for comprehensive energy analyses that include detailed modelling of battery storage systems and their interaction with other energy sources and loads [229]. HOMER's computational demands are relatively low-to-moderate, making it accessible on standard PCs while capable of handling complex hybrid energy system simulations [231]. It can typically run efficiently on systems with a dual-core processor, 8 GB of RAM and an integrated graphics card. Its compatibility with various data formats and integration with other tools make it versatile for different technological environments [232].

4.5. PerModAC

PerModAC, developed by a team at the University of Applied Sciences Berlin (HTW Berlin) in Germany, is a software designed for simulating the performance of PV-BESS systems. This tool integrates various components that are crucial for modelling PV-BESS efficiency, including inverters, batteries, standby functions and energy management system (EMS) control. As of its current version, PerModAC is limited to AC-coupling of BESSs and does not include support for modelling battery ageing [233].

PerModAC is designed to handle various scales of energy projects, from small installations to large industrial applications, including those involving battery energy storage systems [234]. The tool is user-friendly, focusing on ease of use and accessibility for users with different levels of expertise, which is crucial for widespread adoption in both academic and industrial settings. Its interface is intuitive, providing clear guidance for modelling and simulation tasks, making it approachable for new users who need to simulate the performance of battery systems under different operating conditions. This includes evaluating the lifecycles, efficiencies and cost-effectiveness of various battery technologies, such as lithium-ion, lead-acid and flow batteries [234]. PerModAC integrates effectively with other software and data sources, facilitating comprehensive energy analyses and ensuring it can be adapted to different project scales. This integration capability allows users to conduct detailed performance assessments and economic analyses of battery systems in diverse applications, from residential solar-plus-storage setups to large-scale grid stabilisation projects [234]. PerModAC requires moderate computational resources, which can vary depending on the complexity and scale of the energy projects being modelled [234]. Systems with a quad-core processor, 8–16 GB of RAM and a standard graphics card are generally sufficient. It integrates well with other simulation tools and data sources, though users might need to adapt some processes to fully leverage its capabilities [235].

4.6. GridLAB-D

GridLAB-D is used to simulate and analyse power distribution networks. It offers useful data to users who build and manage distribution systems as well as utilities looking to utilise the newest energy technology. The technology is an adaptable simulation environment that has the ability to integrate with various data management and analysis tools from third parties. It was created by Pacific Northwest National Laboratory (PNNL) researchers working with academic and business partners. Its central algorithm coordinates the statuses of multi-component power system networks concurrently, each of which is characterised by a number of differential equations. The following is a comprehensive list of tools that GridLAB-D uses to design, arrange and evaluate studies: agent-based tools, data-driven simulation tools, tools for establishing and authenticating pricing mechanisms, evaluations of customers' responses, verification of how programs communicate and rely on other technologies and retail markets, tools to interface with industry-standard power systems, and analysis mechanisms. These tools can all be used by clients to create comprehensive simulations of the relationships between and the development of new end-use methodologies and distributed energy resources [236]. While the tool is capable of modelling a large-scale micro-grid with a variety of grid constraints, it is devoid of detailed simulation algorithms for energy storage systems and application-specific parameterisation. This limitation renders it unsuitable for in-depth techno-economic analyses and optimisation of storage projects [237].

GridLAB-D is highly scalable and is capable of simulating complex power systems, including large-scale grid operations that incorporate significant amounts of battery storage [236]. Although it is powerful, GridLAB-D can have a steeper learning curve compared to other tools due to its detailed and technical nature, which might require users to have a strong background in power systems engineering and programming [236]. The software interface is less graphical and more command-line based, which may be challenging for some users but offers great flexibility and control over the simulation parameters. This detailed control is particularly beneficial for modelling the dynamic interactions between battery storage systems and other grid components such as generators, loads and renewable energy sources [236]. GridLAB-D integrates well with other power system simulation tools and databases, enhancing its utility in comprehensive grid analyses and adaptability to different project scales. This integration capability is essential for conducting detailed studies on how battery storage can support grid operations, enhance reliability, and facilitate the integration of intermittent renewable energy sources [236]. GridLAB-D is computationally intensive, particularly for large-scale grid simulations, often requiring high-performance computing resources to run efficiently. Optimal performance typically requires a multi-core processor, 16–32 GB of RAM and a dedicated graphics card, along with a robust data management system. It offers strong compatibility with other power system simulation tools and databases, but its command-line interface might present integration challenges for users accustomed to graphical interfaces [238].

4.7. BLAST

The Battery Lifetime Analysis and Simulation Tool (BLAST) suite is a product of the National Renewable Energy Laboratory (NREL) in the United States (US). Proper optimisation systems must be factored in for battery ageing degradation analyses in order to accurately predict the longevity of batteries. BLAST has several variants that are applicable for assessing battery lifespans for behind-the-meter applications (BLAST BTM-Lite), stationary applications (BLAST-S), which allows for the evaluation of battery lifespans and performance in stationary applications, and vehicle applications (BLAST-V), which enables the evaluation of the longevity and the performance of the batteries used for EVs as well for hybrid electric vehicles, plug-in hybrid electric vehicles and battery electric vehicles (BEVs) [239].

BLAST is designed for scalability and is capable of handling both small and largescale energy projects, including those involving battery storage systems [239]. The tool is user-friendly, featuring a clear and intuitive interface that facilitates easy modelling and simulation, making it accessible to users with varying levels of expertise. This ease of use is particularly important for quickly assessing the feasibility and performance of different battery storage solutions in various applications, such as residential solar-plusstorage systems, commercial demand charge management and utility-scale frequency regulation [239]. The graphical interface supports easy interaction and data input, making it accessible for users of various expertise levels and ensuring that complex simulations can be conducted efficiently. BLAST integrates well with other simulation tools and data sources, making it versatile for different applications and adaptable to varying project scales. This integration capability allows users to perform comprehensive evaluations of battery storage systems and to consider factors such as cost, performance and environmental impact [239]. BLAST's computational demands are relatively low, making it suitable for use on standard computer systems while still capable of conducting comprehensive energy simulations. A system with a dual-core processor, 8 GB of RAM, and an integrated graphics card is typically adequate. Its clear and intuitive interface facilitates integration with other tools and data sources, enhancing its adaptability to various technological environments.

4.8. SAM

System Advisor Model (SAM) is a no-cost software package designed to aid decisionmaking among various stakeholders in the renewable energy sector, such as policymakers, project managers, engineers, technology developers and researchers. It allows modelling of a wide array of renewable energy systems encompassing photovoltaic arrays—from small-scale residential setups to extensive utility-level installations—concentrating solar power systems for generating electricity, and wind power systems, ranging from individual turbines to large wind farms. SAM also facilitates battery storage modelling, accommodating various battery types like lithium—ion, lead—acid and flow batteries for different applications—both in front-of-meter and behind-the-meter settings. The software employs detailed capacity, voltage, thermal and lifetime sub-models that are customisable using battery datasheets. SAM's financial tools are flexible for techno—economic evaluations in addition to financial modelling. This includes residential and commercial projects in which a renewable energy system is connected to the electric utility meter on the client's end. This makes it possible to minimise customer billing by utilising the system's power [240,241].

SAM is a robust tool used for estimating the thermal behaviour of Li-ion battery systems, which is crucial for optimising their performance and longevity. SAM incorporates detailed thermal models that consider the impact of various operating conditions on the battery's temperature and degradation [240]. By simulating different thermal management strategies, such as active cooling or passive heat dissipation, SAM offers insights into their effects on battery efficiency and lifespan [242]. Accurate thermal modelling helps prevent issues like thermal runaway, significantly enhancing battery safety and reliability [243]. This capability allows users to develop more efficient and durable battery systems, ensuring optimal performance under diverse conditions [240].

SAM's versatility also extends to simulating both automatic and manual dispatch strategies, which is critical for effective thermal management. Automatic dispatch can be used for applications like peak shaving or responding to time-varying power prices, while manual dispatch allows users to specify charging and discharging schedules [242]. These features help evaluate the thermal impacts of various operational strategies, providing a comprehensive understanding of the trade-offs between enhanced thermal regulation and associated costs [244]. For instance, SAM can simulate the effects of different cooling strategies during peak usage, offering valuable insights into maintaining thermal stability [240,245]. This detailed analysis aids with developing battery systems that are not only efficient but also have extended lifespans, enhancing their overall economic viability [246].

SAM is scalable and is suitable for a wide range of renewable energy projects, from small residential systems to large utility-scale installations, including those that incorporate battery storage [247]. Known for its user-friendly interface and comprehensive support resources, SAM provides extensive tutorials and forums to assist users with designing and evaluating energy systems that include batteries. The software's graphical interface is intuitive, allowing users to easily model and simulate various renewable energy systems, including the integration of battery storage to enhance system reliability and efficiency [247]. SAM integrates effectively with other renewable energy tools and databases, facilitating comprehensive analyses and adaptability to different scales of projects. This integration capability is essential for evaluating the economic viability and performance of battery storage systems, optimizing their configuration and determining their impact on overall system performance [247]. SAM is designed to run efficiently on standard computing hardware, with moderate computational demands even for detailed renewable energy simulations [240]. It can typically operate effectively on systems with a dual-core processor, 8-16 GB of RAM and a standard graphics card. It integrates seamlessly with a variety of renewable energy tools and databases, making it highly adaptable to different technological contexts [247].

4.9. SimSES

Simulation of Stationary Energy Storage Systems (SimSES) is an open-source modelling framework crafted to replicate stationary energy storage systems. Initially created in MATLAB, the tool was later migrated to Python and was further enhanced and collaborated on by a group of researchers at the Technical University of Munich, Germany. SimSES empowers comprehensive assessment and economic modelling of stationary storage systems across different applications. Unique features of this tool include it being a modular, flexible and abstract energy storage model, which enables a user to define the combination of system structure and technology for the energy storage systems [248,249].

In essence, it might be essential to configure and simulate a sufficiently intricate model of a storage system to predict internal conditions like SoC, state-of-health (SOH) or internal losses. Moreover, these simulations should align with the operational requirements specific to the application, ensuring they yield compatible outcomes tailored for a particular setup and forecast the storage system's state accurately. Before proceeding with project acquisition, implementation and operation, it is crucial to conduct comprehensive sensitivity and optimisation analyses using a holistic techno–economic model from the investor's viewpoint. This ensures the most cost-effective integration of storage systems into power grids that are heavily reliant on variable renewable energy sources (vRESs). Additionally, Table 8 offers a quantitative comparison of the technical and economic modelling tools for stationary storage systems discussed in this section. Various parameters affecting deepcycle batteries, such as the temperature, end-of-discharge voltage, discharge current and state-of-health, which significantly impact battery capacity, have also been explored.

SimSES is designed to be scalable, supporting the simulation of both small-scale and large-scale energy storage systems [250]. The tool is user-friendly, with a focus on easeof-use for both novice and experienced users, which is crucial for widespread adoption in academic research and industry. Its interface is intuitive and guides users through the simulation process with clear instructions and visual aids, making it particularly useful for modelling the performance and lifecycles of different battery technologies [250]. SimSES integrates well with other simulation tools and energy management systems, enhancing its applicability in various contexts and its adaptability to different project scales. This integration capability allows users to conduct detailed analyses of battery systems, including their efficiency, degradation and economic impact, providing valuable insights for the design and optimisation of energy storage solutions in both grid-tied and off-grid applications [250]. SimSES requires moderate to high computational resources, particularly for detailed performance and lifecycle simulations of battery storage systems [250]. Optimal performance typically necessitates a multi-core processor, 16 GB of RAM and a dedicated graphics card. It integrates well with other simulation tools and energy management systems, though its specialised focus may require some adaptation for broader applications.

Simulator	Open Source	Web Based	Code Availability	GUI Based	User Manual	Economic Market Analysis	Application	Developer	SoC Estimation
MATLAB	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	MathWorks	\checkmark
StorageVET	\checkmark	\checkmark	×	×	\checkmark	\checkmark	\checkmark	EPRI	\checkmark
DER-VET	\checkmark	×	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	EPRI	\checkmark
HOMER Pro	×	\checkmark	×	\checkmark	\checkmark	\checkmark	×	Homerenergy	×
PerModAC	\checkmark	×	\checkmark	×	\checkmark	\checkmark	×	htw	×
BLAST	×	×	×	\checkmark	\checkmark	\checkmark	\checkmark	NREL	×
GridLab-D	\checkmark	×	×	×	\checkmark	\checkmark	×	PNNL	×
SAM	\checkmark	×	×	\checkmark	\checkmark	\checkmark	×	NREL	×
SimSES	\checkmark	×	\checkmark	×	\checkmark	\checkmark	\checkmark	TUM	\checkmark

Table 8. Characteristics of technical and economic simulation tools for energy storage applications.

5. Conclusions and Future Research Directions

In efforts to replace fuel-based electricity generation with renewable energy sources, implementing energy storage systems has become a promising research area amongst the researcher community to provide stable, available and reliable energy with respect to supply and demand. Battery management systems have become the preferred energy storage system due to their high power density and low self-discharging. A comprehensive analysis and evaluation of energy storage technologies, particularly focusing on electrochemical and battery-based storage, is presented. The discussion emphasises the role of battery energy storage systems in storing energy for photovoltaic (PV) applications, highlighting the

diverse characteristics of the batteries used in these setups. Various methods for estimating the SoC are explored and are categorised into different groups, each possessing unique attributes. Additionally, a concise comparison and assessment of different technical and economic modelling tools for energy storage applications is outlined, considering that no single tool suits every application scenario.

Renewable energy sources exhibit significant variability depending on the location. For instance, solar energy availability heavily relies on the latitude and climate conditions. Furthermore, even within a specific location, renewable resources can fluctuate significantly on a seasonal and hourly basis. The behaviour and economic feasibility of renewable power systems are directly influenced by the nature of available resources, impacting the quantity and timing of renewable power generation. Therefore, accurately modelling renewable resources is crucial in system analysis and planning. Renewable energy sources will continue to be more pervasive in the energy sector and are envisaged to positively provide an eco-friendlier, reliable and sustainable power supply, especially in austere and remote locations where the possibility of national-grid electrification is always a major issue to be addressed. The intermittent nature of many renewable energy sources is challenging to the daily operation and maintenance of the electric grid. Renewable intermittency refers to unforeseen fluctuations and variations in output power from wind and solar sources. This erratic output power of wind and solar power, which are subject to wind speed and solar irradiance variations, affect power system reliability and the stability in the distribution network. However, microgrids are an enabling technology for utilising renewable energy sources. The inherent intermittency and the variability associated with renewable energy sources become more complicated for microgrid operations. Since renewable energy sources (solar irradiance and wind speed, which are the major consequences of the volatile nature of wind and solar energy) will continue to disrupt the operation of the grid as a result of intermittency and uncertainty, it is critical to take into consideration an efficient energy storage system to store energy for future use.

Simulation and modelling tools for energy storage cannot universally cover all the essentials for designing, modelling and analysing systems due to the specificity of each application's needs. Consequently, it is crucial to scrutinise system approaches meticulously, aligning them with the mechanisms and objectives of the particular application in question. Applications vary in terms of scale, technology requirements and operational parameters, demanding tailored approaches to effectively capture their nuances and to address their unique challenges. This necessitates a thoughtful evaluation of methodologies to ensure they align with the intricacies and goals of the given use-case, thus enhancing their applicability and relevance.

SoC estimation is seen as an important parameter in every application related to energy storage systems. This is because it assists with monitoring batteries and protecting them from deep-charging and overcharging. Having an idea of the actual SoC gives proper information on the real-time energy status of batteries, thereby giving assurance of a reliable operation process. From the study, it is observed and evident that some of the SoC estimation approaches, like the neural network technique, though effective, require a lot of training data for accuracy. Also, network convergence is slow, making it difficult to achieve the global optimum. These issues are also similar with the different SoC estimation approaches discussed in this study. Therefore, it is recommended that novel approaches are modelled and implemented to solve the present challenges of these existing SoC estimating methods in confronting PV application use-cases for reliable and sustainable power demand in austere and remote locations.

Despite advancements in SoC estimation techniques, certain gaps persist. One notable area is the need for improved accuracy, especially under diverse operating conditions. Current methods might struggle with accurately estimating the SoC in extreme temperatures, varied charge/discharge rates or with aged batteries. There is a necessity for robust algorithms that can adapt to these conditions more effectively. Additionally, achieving real-time estimation methods that are both precise and computationally efficient remain

an ongoing challenge. Balancing accuracy with computational complexity is crucial for practical implementation, yet achieving this balance is an area that requires further exploration. Furthermore, the development of standardised validation methodologies for comparing and benchmarking SoC estimation techniques across various battery chemistries and system configurations is an area that requires attention.

In the realm of energy storage technologies, several gaps exist. While lithium–ion batteries dominate the market, their limitations in terms of safety, lifespan and environmental impact highlight the need for alternative technologies. Research in this area often focuses on enhancing the performance and safety of existing technologies. However, there is a gap in the development of entirely new, disruptive technologies that could potentially surpass the limitations of current options. Additionally, there is a need for scalability and cost-effectiveness, especially for grid-scale applications. Finding solutions that balance performance, scalability and cost across different energy storage technologies remains a challenge. Exploring novel materials, alternative chemistries and innovative storage concepts could address these gaps.

Current simulators for energy storage systems often lack comprehensive integration of various components, hindering their accuracy and applicability. There is a gap in creating simulation platforms that accurately capture the complex interplay between energy storage components, control strategies and external factors like grid dynamics or variable renewable energy sources. Integration of real-time data and control algorithms into simulation models for accurate real-world scenarios is also an area needing attention. Moreover, there is a lack of standardised and open-source simulation platforms that can facilitate collaboration and benchmarking across the research community. Developing such platforms would enable researchers to validate and compare their models, fostering innovation and improvement within the field.

Addressing these research gaps will significantly contribute to advancing the reliability, efficiency and applicability of SoC estimation methods, energy storage technologies and state-of-the-art simulators, which are crucial for the future development and deployment of efficient energy storage systems.

Author Contributions: Conceptualization, T.K., H.C.M. and A.D.F.; Data curation, T.K.; Funding acquisition, H.C.M.; Investigation, T.K.; Methodology, T.K., H.C.M. and A.D.F.; Project administration, H.C.M. and A.D.F.; Resources, H.C.M.; Supervision, H.C.M.; Writing—original draft, T.K.; Writing—review and editing, T.K., H.C.M. and A.D.F. All authors have read and agreed to the published version of the manuscript.

Funding: This research was sponsored by the ARO and was accomplished under grant number: W911NF-22-1-0006. The views and conclusions contained in this document are those of the authors and should not be interpreted as representing the official policies, either expressed or implied, of the ARO or the US government. The US government is authorized to reproduce and distribute reprints for government purposes notwithstanding any copyright notation herein.

Acknowledgments: The US Army, Department of Defence, provided financial support for this research.

Conflicts of Interest: The authors declare no conflicts of interest.

Abbreviations

The following abbreviations are used in this article:

Alternating Current
Absorbent Glass Mat
Artificial Intelligence
Ampere-Hours
Battery Energy Storage System
Battery Management System
Battery Storage System
Cadmium
Coulomb Counting

CPCS	Control and Power Conditioning System
DoD	Depth of Discharge
DC	Direct Current
DERVET	Distributed Energy Value Estimation Tool
EAP	Electrode Ageing Parameter
EMF	Electromotive Force
EV	Electric Vehicle
EKF	Extended Kalman Filter
HEMS	Home Energy Management System
HOMER	Hybrid Optimisation Model for Multiple Energy Resources
KF	Kalman Filter
KOH	Alkaline Potassium Hydroxide
LFP	Lithium Iron Phosphate
Li-ion	Lithium–Ion
LM	Levenberg–Marquardt
LSA	Lightning Search Algorithm
LMO	Lithium Manganese Oxide
LTO	Lithium Titanate
ML	Machine Learning
NN	Neural Network
NMC	Lithium Nickel Manganese Cobalt Oxide
NaS	Sodium Sulfur
NiCd	Nickel–Cadmium
OCV	Open-Circuit Voltage
PbA	Lead–Acid
PV	Photovoltaic
RMSE	Root Mean Square Error
RV	Recreational Vehicle
SCG	Scaled Conjugate Gradient
SLA	Sealed Lead–Acid
SoC	State-of-Charge
SoH	Sate-of-Heath
SoL	State-of-Life
SoP	State-of-Power
VRLA	Valve-Regulated Lead–Acid

References

- 1. Qays, M.O.; Buswig, Y.; Hossain, M.L.; Abu-Siada, A. Recent progress and future trends on the state of charge estimation methods to improve battery-storage efficiency: A review. *CSEE J. Power Energy Syst.* **2020**, *8*, 105–114.
- 2. Xiong, R.; Cao, J.; Yu, Q.; He, H.; Sun, F. Critical review on the battery state of charge estimation methods for electric vehicles. *IEEE Access* 2017, *6*, 1832–1843. [CrossRef]
- 3. Khalid, A.; Sarwat, A.I. Unified univariate-neural network models for lithium-ion battery state-of-charge forecasting using minimized akaike information criterion algorithm. *IEEE Access* **2021**, *9*, 39154–39170. [CrossRef]
- 4. Loukil, J.; Masmoudi, F.; Derbel, N. A real-time estimator for model parameters and state of charge of lead acid batteries in photovoltaic applications. *J. Energy Storage* **2021**, *34*, 102184. [CrossRef]
- Harigopal, A.; Nithin, S. Assessment of State of Charge estimation techniques for Li-Ion battery pack. In Proceedings of the 2020 International Conference on Smart Electronics and Communication (ICOSEC), Trichy, India, 10–12 September 2020; pp. 988–991.
- 6. Hannan, M.A.; Lipu, M.; Hussain, A.; Ker, P.J.; Mahlia, T.I.; Mansor, M.; Ayob, A.; Saad, M.H.; Dong, Z. Toward enhanced state of charge estimation of lithium-ion batteries using optimized machine learning techniques. *Sci. Rep.* **2020**, *10*, 4687. [CrossRef]
- Shah, A.; Shah, K.; Shah, C.; Shah, M. State of charge, remaining useful life and knee point estimation based on artificial intelligence and Machine learning in lithium-ion EV batteries: A comprehensive review. *Renew. Energy Focus* 2022, 42, 146–164. [CrossRef]
- 8. Shen, M.; Gao, Q. A review on battery management system from the modeling efforts to its multiapplication and integration. *Int. J. Energy Res.* **2019**, *43*, 5042–5075. [CrossRef]
- Ipek, E.; Eren, M.K.; Yilmaz, M. State-of-charge estimation of li-ion battery cell using support vector regression and gradient boosting techniques. In Proceedings of the 2019 International Aegean Conference on Electrical Machines and Power Electronics (ACEMP) & 2019 International Conference on Optimization of Electrical and Electronic Equipment (OPTIM), Istanbul, Turkey, 27–29 August 2019; pp. 604–609.

- 10. Girijaprasanna, T.; Dhanamjayulu, C. A Review on Different State of Battery Charge Estimation Techniques and Management Systems for EV Applications. *Electronics* **2022**, *11*, 1795. [CrossRef]
- Khan, M.A.A.; Khalid, H.A.; Balan, R.; Bakkaloglu, B. A novel State of Charge and State of Health estimation technique for Lithium-ion cells using machine learning based Pseudo-Random Binary Sequence method. *J. Energy Storage* 2022, 55, 105472. [CrossRef]
- Varshney, A.; Singh, A.; Pradeep, A.A.; Joseph, A.; Gopakumar, P. Monitoring State of Health and State of Charge of Lithium-Ion Batteries Using Machine Learning Techniques. In Proceedings of the 2021 IEEE 5th International Conference on Condition Assessment Techniques in Electrical Systems (CATCON), Kozhikode, India, 3–5 December 2021; pp. 22–27.
- 13. Hemavathi, S. Modeling and Estimation of Lithium-ion Battery State of Charge Using Intelligent Techniques. In *Advances in Power and Control Engineering*; Springer: Berlin/Heidelberg, Germany, 2020; pp. 157–172.
- 14. Narayanan, S.S.S.; Thangavel, S. Machine learning-based model development for battery state of charge–open circuit voltage relationship using regression techniques. *J. Energy Storage* **2022**, *49*, 104098. [CrossRef]
- 15. Li, J.; He, S.; Yang, Q.; Wei, Z.; Li, Y.; He, H. A Comprehensive Review of Second Life Batteries Towards Sustainable Mechanisms: Potential, Challenges, and Future Prospects. *IEEE Trans. Transp. Electrif.* **2022**, *9*, 4824–4845. [CrossRef]
- Shahjalal, M.; Roy, P.K.; Shams, T.; Fly, A.; Chowdhury, J.I.; Ahmed, M.R.; Liu, K. A review on second-life of Li-ion batteries: Prospects, challenges, and issues. *Energy* 2022, 241, 122881. [CrossRef]
- 17. McDougall, N. The Operational Environment for Repurposing Electric Vehicle Lithium-Ion Batteries for Energy Storage Applications in the EU. Master's Thesis, Aalto University, Espoo, Finland, 2023.
- 18. Xu, L.; Lei, S.; Srinivasan, D.; Song, Z. Can retired lithium-ion batteries be a game changer in fast charging stations? *eTransportation* **2023**, *18*, 100297. [CrossRef]
- 19. Hua, Y.; Liu, X.; Zhou, S.; Huang, Y.; Ling, H.; Yang, S. Toward sustainable reuse of retired lithium-ion batteries from electric vehicles. *Resour. Conserv. Recycl.* 2021, *168*, 105249. [CrossRef]
- 20. Chombo, P.V.; Laoonual, Y. A review of safety strategies of a Li-ion battery. J. Power Sources 2020, 478, 228649. [CrossRef]
- Ogunfuye, S.A. Lithium-Ion Battery Safety Analysis with Physical Sub-Models. Doctoral Dissertation, West Virginia University, Morgantown, WV, USA, 2023.
- 22. Faraji Niri, M.; Aslansefat, K.; Haghi, S.; Hashemian, M.; Daub, R.; Marco, J. A Review of the Applications of Explainable Machine Learning for Lithium–Ion Batteries: From Production to State and Performance Estimation. *Energies* 2023, *16*, 6360. [CrossRef]
- 23. Du, B.; Yu, Z.; Yi, S.; He, Y.; Luo, Y. State-of-charge estimation for second-life lithium-ion batteries based on cell difference model and adaptive fading unscented Kalman filter algorithm. *Int. J. Low-Carbon Technol.* **2021**, *16*, 927–939. [CrossRef]
- Luo, Y.F. A multi-frequency electrical impedance spectroscopy technique of artificial neural network-based for the static state of charge. *Energies* 2021, 14, 2526. [CrossRef]
- Dao, V.Q.; Dinh, M.C.; Kim, C.S.; Park, M.; Doh, C.H.; Bae, J.H.; Lee, M.K.; Liu, J.; Bai, Z. Design of an effective State of Charge estimation method for a lithium-ion battery pack using extended kalman filter and artificial neural network. *Energies* 2021, 14, 2634. [CrossRef]
- Dewalkar, S.; Nangrani, S. Artificial Intelligence-Based State of Charge Estimation of Electric Vehicle Battery. In Smart Technologies for Energy, Environment and Sustainable Development, Vol 2; Springer: Berlin/Heidelberg, Germany, 2022; pp. 699–705.
- 27. Abraham, T.R.; Sunil, K.; Shah, M.; Ashok, N.; Thomas, S. Energy storage devices: Batteries and supercapacitors. In *Nanobiohybrids* for Advanced Wastewater Treatment and Energy Recovery; IWA Publishing: London, UK, 2023; pp. 61–84.
- 28. Tarascon, J.M.; Armand, M. Issues and challenges facing rechargeable lithium batteries. Nature 2001, 414, 359–367. [CrossRef]
- 29. Alqahtani, H.; Kumar, G. Machine learning for enhancing transportation security: A comprehensive analysis of electric and flying vehicle systems. *Eng. Appl. Artif. Intell.* **2024**, 129, 107667. [CrossRef]
- 30. Mishra, P.; Singh, G. Energy management systems in sustainable smart cities based on the Internet of energy: A technical review. *Energies* **2023**, *16*, 6903. [CrossRef]
- Naraindath, N.R.; Kupolati, H.A.; Bansal, R.C.; Naidoo, R.M. Data security and privacy, cyber-security enhancement, and systems recovery approaches for microgrid networks. In *Modelling and Control Dynamics in Microgrid Systems with Renewable Energy Resources*; Elsevier: Amsterdam, The Netherlands, 2024; pp. 377–401.
- 32. Divya, K.; Østergaard, J. Battery energy storage technology for power systems—An overview. *Electr. Power Syst. Res.* 2009, 79, 511–520. [CrossRef]
- Palizban, O.; Kauhaniemi, K. Energy storage systems in modern grids—Matrix of technologies and applications. J. Energy Storage 2016, 6, 248–259. [CrossRef]
- Koohi-Fayegh, S.; Rosen, M.A. A review of energy storage types, applications and recent developments. J. Energy Storage 2020, 27, 101047. [CrossRef]
- Díaz-González, F.; Sumper, A.; Gomis-Bellmunt, O.; Villafáfila-Robles, R. A review of energy storage technologies for wind power applications. *Renew. Sustain. Energy Rev.* 2012, 16, 2154–2171. [CrossRef]
- Bhatia, A. Design and Sizing of Solar Photovoltaic Systems; Continuing Education and Development Inc.: Woodcliff Lake, NJ, USA, 2022; pp. 2–125. Available online: https://www.cedengineering.com/courses/design-and-sizing-of-solar-photovoltaic-systems (accessed on 6 August 2024).
- BU-201: How Does the Lead Acid Battery Work? Available online: https://batteryuniversity.com/article/bu-201-how-does-thelead-acid-battery-work (accessed on 24 August 2022).

- 38. BU-202: New Lead Acid Systems. Available online: https://batteryuniversity.com/article/bu-202-new-lead-acid-systems (accessed on 24 August 2022).
- Olabi, A.G.; Abbas, Q.; Shinde, P.A.; Abdelkareem, M.A. Rechargeable batteries: Technological advancement, challenges, current and emerging applications. *Energy* 2023, 266, 126408. [CrossRef]
- 40. Bindner, H.; Cronin, T.; Lundsager, P.; Manwell, J.F.; Abdulwahid, U.; Baring-Gould, I. *Lifetime Modelling of Lead acid Batteries*; Elsevier: Amsterdam, The Netherlands, 2005.
- 41. Relion Battery. Lithium Battery Depth of Discharge, State of Charge, and the Affect on Battery Capacity. Available online: https://www.relionbattery.com/ (accessed on 10 August 2024).
- 42. Kiessling, R. Lead Acid Battery Formation Techniques; Digatron Firing Circuits, Digatron: Aachen, Germany, 1992; p. 2.
- 43. Mandal, S.; Thangarasu, S.; Thong, P.T.; Kim, S.C.; Shim, J.Y.; Jung, H.Y. Positive electrode active material development opportunities through carbon addition in the lead-acid batteries: A recent progress. *J. Power Sources* **2021**, *485*, 229336. [CrossRef]
- 44. BU-203: Nickel-Based Batteries. Available online: https://batteryuniversity.com/article/bu-203-nickel-based-batteries (accessed on 24 August 2022).
- Patel, M.; Mishra, K.; Banerjee, R.; Chaudhari, J.; Kanchan, D.; Kumar, D. Fundamentals, recent developments and prospects of lithium and non-lithium electrochemical rechargeable battery systems. *J. Energy Chem.* 2023, *81*, 221–259. [CrossRef]
- 46. Petrovic, S.; Petrovic, S. Nickel—Cadmium batteries. In *Battery Technology Crash Course: A Concise Introduction;* Springer: Berlin/Heidelberg, Germany, 2021; pp. 73–88.
- 47. Wen, Z.; Cao, J.; Gu, Z.; Xu, X.; Zhang, F.; Lin, Z. Research on sodium sulfur battery for energy storage. *Solid State Ionics* 2008, 179, 1697–1701. [CrossRef]
- 48. Handbook on Battery Energy Storage System. 2018. Available online: https://www.adb.org/sites/default/files/publication/47 9891/handbook-battery-energy-storage-system.pdf (accessed on 24 August 2022).
- 49. Broussely, M.; Pistoia, G. Industrial Applications of Batteries: From Cars to Aerospace and Energy Storage; Elsevier: Amsterdam, The Netherlands, 2007.
- Wang, Y.; Zhou, D.; Palomares, V.; Shanmukaraj, D.; Sun, B.; Tang, X.; Wang, C.; Armand, M.; Rojo, T.; Wang, G. Revitalising sodium–sulfur batteries for non-high-temperature operation: A crucial review. *Energy Environ. Sci.* 2020, 13, 3848–3879. [CrossRef]
- 51. Palomares, V.; Hueso, K.B.; Armand, M.; Rojo, T. High-Temperature Battery Technologies: Na-S. *Batteries: Present and Future Energy Storage Challenges*; Wiley: Hoboken, NJ, USA, 2020; p. 371.
- 52. Nikiforidis, G.; Jongerden, G.; Jongerden, E.; Van De Sanden, M.; Tsampas, M. An electrochemical study on the cathode of the intermediate temperature tubular sodium-sulfur (NaS) battery. *J. Electrochem. Soc.* **2019**, *166*, A135. [CrossRef]
- 53. Habib, A.R.R.; Butler, K. Environmental and economic comparison of hydrogen fuel cell and battery electric vehicles. *Future Technol.* **2022**, *1*, 25–33. [CrossRef]
- Ding, Y.; Cano, Z.P.; Yu, A.; Lu, J.; Chen, Z. Automotive Li-ion batteries: Current status and future perspectives. *Electrochem. Energy Rev.* 2019, 2, 1–28. [CrossRef]
- 55. Bubulinca, C.; Kazantseva, N.E.; Pechancova, V.; Joseph, N.; Fei, H.; Venher, M.; Ivanichenko, A.; Saha, P. Development of All-Solid-State Li-Ion Batteries: From Key Technical Areas to Commercial Use. *Batteries* **2023**, *9*, 157. [CrossRef]
- 56. Randau, S.; Weber, D.A.; Kötz, O.; Koerver, R.; Braun, P.; Weber, A.; Ivers-Tiffée, E.; Adermann, T.; Kulisch, J.; Zeier, W.G.; et al. Benchmarking the performance of all-solid-state lithium batteries. *Nat. Energy* **2020**, *5*, 259–270. [CrossRef]
- 57. Manthiram, A.; Yu, X.; Wang, S. Lithium battery chemistries enabled by solid-state electrolytes. *Nat. Rev. Mater.* **2017**, *2*, 1–16. [CrossRef]
- 58. Rahman, M.A.; Song, G.; Bhatt, A.I.; Wong, Y.C.; Wen, C. Nanostructured silicon anodes for high-performance lithium-ion batteries. *Adv. Funct. Mater.* **2016**, *26*, 647–678. [CrossRef]
- 59. Wu, J.; Cao, Y.; Zhao, H.; Mao, J.; Guo, Z. The critical role of carbon in marrying silicon and graphite anodes for high-energy lithium-ion batteries. *Carbon Energy* **2019**, *1*, 57–76. [CrossRef]
- Xu, Z.L.; Liu, X.; Luo, Y.; Zhou, L.; Kim, J.K. Nanosilicon anodes for high performance rechargeable batteries. *Prog. Mater. Sci.* 2017, 90, 1–44. [CrossRef]
- 61. Zhao, Y.; Ye, Y.; Wu, F.; Li, Y.; Li, L.; Chen, R. Anode interface engineering and architecture design for high-performance lithium–sulfur batteries. *Adv. Mater.* **2019**, *31*, 1806532. [CrossRef]
- 62. Ghalkhani, M.; Habibi, S. Review of the Li-ion battery, thermal management, and AI-based battery management system for EV application. *Energies* 2022, *16*, 185. [CrossRef]
- 63. Zhao, J.; Feng, X.; Tran, M.K.; Fowler, M.; Ouyang, M.; Burke, A.F. Battery safety: Fault diagnosis from laboratory to real world. *J. Power Sources* **2024**, *598*, 234111. [CrossRef]
- 64. Yang, Y. A machine-learning prediction method of lithium-ion battery life based on charge process for different applications. *Applied Energy* **2021**, 292, 116897. [CrossRef]
- 65. Liu, K.; Li, K.; Peng, Q.; Zhang, C. A brief review on key technologies in the battery management system of electric vehicles. *Front. Mech. Eng.* **2019**, *14*, 47–64. [CrossRef]
- 66. Atalay, S.; Sheikh, M.; Mariani, A.; Merla, Y.; Bower, E.; Widanage, W.D. Theory of battery ageing in a lithium-ion battery: Capacity fade, nonlinear ageing and lifetime prediction. *J. Power Sources* **2020**, *478*, 229026. [CrossRef]
- 67. Shchurov, N.I.; Dedov, S.I.; Malozyomov, B.V.; Shtang, A.A.; Martyushev, N.V.; Klyuev, R.V.; Andriashin, S.N. Degradation of lithium-ion batteries in an electric transport complex. *Energies* **2021**, *14*, 8072. [CrossRef]

- 68. Yang, S.; Zhang, C.; Jiang, J.; Zhang, W.; Gao, Y.; Zhang, L. A voltage reconstruction model based on partial charging curve for state-of-health estimation of lithium-ion batteries. *J. Energy Storage* **2021**, *35*, 102271. [CrossRef]
- 69. Zheng, Y.; Ouyang, M.; Han, X.; Lu, L.; Li, J. Investigating the error sources of the online state of charge estimation methods for lithium-ion batteries in electric vehicles. *J. Power Sources* 2018, 377, 161–188. [CrossRef]
- 70. Tanim, T.R.; Rahn, C.D.; Wang, C.Y. State of charge estimation of a lithium ion cell based on a temperature dependent and electrolyte enhanced single particle model. *Energy* **2015**, *80*, 731–739. [CrossRef]
- Carrasco Ortega, P.; Durán Gómez, P.; Mérida Sánchez, J.C.; Echevarría Camarero, F.; Pardiñas, Á.Á. Battery energy storage systems for the new electricity market landscape: Modeling, state diagnostics, management, and viability—A review. *Energies* 2023, 16, 6334. [CrossRef]
- 72. Zhang, D.; Park, S.; Couto, L.D.; Viswanathan, V.; Moura, S.J. Beyond battery state of charge estimation: Observer for electrodelevel state and cyclable lithium with electrolyte dynamics. *IEEE Trans. Transp. Electrif.* 2022, *9*, 4846–4861. [CrossRef]
- 73. Chang, W.Y. The state of charge estimating methods for battery: A review. Int. Sch. Res. Not. 2013, 2013, 953792. [CrossRef]
- 74. Tennyson, E.M.; Garrett, J.L.; Frantz, J.A.; Myers, J.D.; Bekele, R.Y.; Sanghera, J.S.; Munday, J.N.; Leite, M.S. Nanoimaging of open-circuit voltage in photovoltaic devices. *Adv. Energy Mater.* 2015, *5*, 1501142. [CrossRef]
- Ali, M.U.; Zafar, A.; Nengroo, S.H.; Hussain, S.; Junaid Alvi, M.; Kim, H.J. Towards a smarter battery management system for electric vehicle applications: A critical review of lithium-ion battery state of charge estimation. *Energies* 2019, 12, 446. [CrossRef]
- 76. Qiao, J.; Wang, S.; Yu, C.; Shi, W.; Fernandez, C. A novel bias compensation recursive least square-multiple weighted dual extended Kalman filtering method for accurate state-of-charge and state-of-health co-estimation of lithium-ion batteries. *Int. J. Circuit Theory Appl.* 2021, 49, 3879–3893. [CrossRef]
- 77. Jeon, S.; Yun, J.J.; Bae, S. Comparative study on the battery state-of-charge estimation method. *Indian J. Sci. Technol.* **2015**, *8*, 1–6. [CrossRef]
- 78. Xiong, R.; Tian, J.; Shen, W.; Sun, F. A novel fractional order model for state of charge estimation in lithium ion batteries. *IEEE Trans. Veh. Technol.* **2018**, *68*, 4130–4139. [CrossRef]
- 79. Jiang, C.; Wang, S.; Wu, B.; Fernandez, C.; Xiong, X.; Coffie-Ken, J. A state-of-charge estimation method of the power lithium-ion battery in complex conditions based on adaptive square root extended Kalman filter. *Energy* **2021**, *219*, 119603. [CrossRef]
- 80. Zhang, R.; Xia, B.; Li, B.; Cao, L.; Lai, Y.; Zheng, W.; Wang, H.; Wang, W.; Wang, M. A study on the open circuit voltage and state of charge characterization of high capacity lithium-ion battery under different temperature. *Energies* **2018**, *11*, 2408. [CrossRef]
- 81. Chen, X.; Lei, H.; Xiong, R.; Shen, W.; Yang, R. A novel approach to reconstruct open circuit voltage for state of charge estimation of lithium ion batteries in electric vehicles. *Appl. Energy* **2019**, 255, 113758. [CrossRef]
- 82. Gismero, A.; Schaltz, E.; Stroe, D.I. Recursive state of charge and state of health estimation method for lithium-ion batteries based on coulomb counting and open circuit voltage. *Energies* **2020**, *13*, 1811. [CrossRef]
- Tian, J.; Xiong, R.; Shen, W.; Sun, F. Electrode ageing estimation and open circuit voltage reconstruction for lithium ion batteries. Energy Storage Mater. 2021, 37, 283–295. [CrossRef]
- 84. Dang, X.; Yan, L.; Xu, K.; Wu, X.; Jiang, H.; Sun, H. Open-circuit voltage-based state of charge estimation of lithium-ion battery using dual neural network fusion battery model. *Electrochim. Acta* 2016, *188*, 356–366. [CrossRef]
- Sato, S.; Kawamura, A. A new estimation method of state of charge using terminal voltage and internal resistance for lead acid battery. In Proceedings of the Power Conversion Conference-Osaka 2002 (Cat. No. 02TH8579), Osaka, Japan, 2–5 April 2002; Volume 2, pp. 565–570.
- Zheng, L.; Zhu, J.; Lu, D.D.C.; Wang, G.; He, T. Incremental capacity analysis and differential voltage analysis based state of charge and capacity estimation for lithium-ion batteries. *Energy* 2018, 150, 759–769. [CrossRef]
- 87. Ren, H.; Zhao, Y.; Chen, S.; Wang, T. Design and implementation of a battery management system with active charge balance based on the SOC and SOH online estimation. *Energy* **2019**, *166*, 908–917. [CrossRef]
- 88. Kim, J.; Kowal, J. A Method for Monitoring State-of-Charge of Lithium-Ion Cells Using Multi-Sine Signal Excitation. *Batteries* **2021**, *7*, 76. [CrossRef]
- 89. Liu, C.; Li, Q.; Wang, K. State-of-charge estimation and remaining useful life prediction of supercapacitors. *Renew. Sustain. Energy Rev.* 2021, 150, 111408. [CrossRef]
- Rezaei, O.; Moghaddam, H.A.; Papari, B. A fast sliding-mode-based estimation of state-of-charge for lithium-ion batteries for electric vehicle applications. J. Energy Storage 2022, 45, 103484. [CrossRef]
- Yang, B.; Wang, J.; Cao, P.; Zhu, T.; Shu, H.; Chen, J.; Zhang, J.; Zhu, J. Classification, summarization and perspectives on state-of-charge estimation of lithium-ion batteries used in electric vehicles: A critical comprehensive survey. *J. Energy Storage* 2021, 39, 102572. [CrossRef]
- 92. Zhang, J.; Wang, P.; Liu, Y.; Cheng, Z. Variable-order equivalent circuit modeling and state of charge estimation of lithium-ion battery based on electrochemical impedance spectroscopy. *Energies* **2021**, *14*, 769. [CrossRef]
- Vadhva, P.; Hu, J.; Johnson, M.J.; Stocker, R.; Braglia, M.; Brett, D.J.; Rettie, A.J. Electrochemical Impedance Spectroscopy for All-Solid-State Batteries: Theory, Methods and Future Outlook. *ChemElectroChem* 2021, *8*, 1930–1947. [CrossRef]
- Zhu, X.; Hallemans, N.; Wouters, B.; Claessens, R.; Lataire, J.; Hubin, A. Operando odd random phase electrochemical impedance spectroscopy as a promising tool for monitoring lithium-ion batteries during fast charging. *J. Power Sources* 2022, 544, 231852. [CrossRef]

- 95. Ruan, H.; Sun, B.; Jiang, J.; Zhang, W.; He, X.; Su, X.; Bian, J.; Gao, W. A modified-electrochemical impedance spectroscopy-based multi-time-scale fractional-order model for lithium-ion batteries. *Electrochim. Acta* **2021**, *394*, 139066. [CrossRef]
- Zhang, X.; Zhang, X.; Sun, X.; An, Y.; Song, S.; Li, C.; Wang, K.; Su, F.; Chen, C.M.; Liu, F.; et al. Electrochemical impedance spectroscopy study of lithium-ion capacitors: Modeling and capacity fading mechanism. *J. Power Sources* 2021, 488, 229454. [CrossRef]
- 97. Xu, J.; Mi, C.C.; Cao, B.; Cao, J. A new method to estimate the state of charge of lithium-ion batteries based on the battery impedance model. *J. Power Sources* 2013, 233, 277–284. [CrossRef]
- Zhang, Q.; Huang, C.G.; Li, H.; Feng, G.; Peng, W. Electrochemical impedance spectroscopy based state of health estimation for lithium-ion battery considering temperature and state of charge effect. *IEEE Trans. Transp. Electrif.* 2022, *8*, 4633–4645. [CrossRef]
- 99. Mc Carthy, K.; Gullapalli, H.; Ryan, K.M.; Kennedy, T. Electrochemical impedance correlation analysis for the estimation of Li-ion battery state of charge, state of health and internal temperature. *J. Energy Storage* **2022**, *50*, 104608. [CrossRef]
- 100. Guo, Y.; Yang, Z.; Liu, K.; Zhang, Y.; Feng, W. A compact and optimized neural network approach for battery state-of-charge estimation of energy storage system. *Energy* **2021**, *219*, 119529. [CrossRef]
- 101. Movassagh, K.; Raihan, A.; Balasingam, B.; Pattipati, K. A critical look at coulomb counting approach for state of charge estimation in batteries. *Energies* **2021**, *14*, 4074. [CrossRef]
- Ko, Y.; Cho, K.; Kim, M.; Choi, W. A Novel Capacity Estimation Method for the Lithium Batteries Using the Enhanced Coulomb Counting Method With Kalman Filtering. *IEEE Access* 2022, 10, 38793–38801. [CrossRef]
- Li, X.; Xiao, L.; Geng, G.; Jiang, Q. Temperature characterization based state-of-charge estimation for pouch lithium-ion battery. J. Power Sources 2022, 535, 231441. [CrossRef]
- Danko, M.; Adamec, J.; Taraba, M.; Drgona, P. Overview of batteries State of Charge estimation methods. *Transp. Res. Procedia* 2019, 40, 186–192. [CrossRef]
- 105. Baccouche, I.; Jemmali, S.; Mlayah, A.; Manai, B.; Amara, N.E.B. Implementation of an improved Coulomb-counting algorithm based on a piecewise SOC-OCV relationship for SOC estimation of li-IonBattery. *arXiv* **2018**, arXiv:1803.10654.
- Movassagh, K.; Raihan, S.A.; Balasingam, B. Performance analysis of coulomb counting approach for state of charge estimation. In Proceedings of the 2019 IEEE Electrical Power and Energy Conference (EPEC), Montreal, QC, Canada, 16–18 October 2019; pp. 1–6.
- 107. Zhang, S.; Li, J.; Li, R.; Zhang, X. Voltage sensor fault detection, isolation and estimation for lithium-ion battery used in electric vehicles via a simple and practical method. *J. Energy Storage* 2022, *55*, 105555. [CrossRef]
- Pop, V.; Bergveld, H.J.; Notten, P.; Regtien, P.P. State-of-the-art of battery state-of-charge determination. *Meas. Sci. Technol.* 2005, 16, R93. [CrossRef]
- 109. Qiao, X.; Wang, Z.; Hou, E.; Liu, G.; Cai, Y. Online estimation of open circuit voltage based on extended kalman filter with self-evaluation criterion. *Energies* 2022, 15, 4373. [CrossRef]
- 110. Rivera-Barrera, J.P.; Muñoz-Galeano, N.; Sarmiento-Maldonado, H.O. SoC estimation for lithium-ion batteries: Review and future challenges. *Electronics* **2017**, *6*, 102. [CrossRef]
- Rodrigues, L.M.; Montez, C.; Moraes, R.; Portugal, P.; Vasques, F. A temperature-dependent battery model for wireless sensor networks. *Sensors* 2017, 17, 422. [CrossRef]
- 112. Talluri, M.T.; Murugesan, S.; Karthikeyan, V.; Pragaspathy, S. The capacity estimation of Li–Ion battery using ML-based hybrid model. *Electr. Eng.* **2024**, 1–11.. [CrossRef]
- Sindhuja, S.; Vasanth, K. Modified coulomb counting method of SOC estimation for uninterruptible power supply system's battery management system. In Proceedings of the 2015 International Conference on Control, Instrumentation, Communication and Computational Technologies (ICCICCT), Kumaracoil, India, 18–19 December 2015; pp. 197–203.
- Zhang, S.; Guo, X.; Dou, X.; Zhang, X. A data-driven coulomb counting method for state of charge calibration and estimation of lithium-ion battery. *Sustain. Energy Technol. Assess.* 2020, 40, 100752. [CrossRef]
- 115. Lei, Z.; Liu, T.; Sun, X.; Xie, H.; Sun, Q. Extended state observer assisted Coulomb counting method for battery state of charge estimation. *Int. J. Energy Res.* 2021, 45, 3157–3169. [CrossRef]
- 116. Vidal, C.; Malysz, P.; Kollmeyer, P.; Emadi, A. Machine learning applied to electrified vehicle battery state of charge and state of health estimation: State-of-the-art. *IEEE Access* 2020, *8*, 52796–52814. [CrossRef]
- 117. Lipu, M.S.H.; Hannan, M.A.; Hussain, A.; Saad, M.H.; Ayob, A.; Uddin, M.N. Extreme learning machine model for state-of-charge estimation of lithium-ion battery using gravitational search algorithm. *IEEE Trans. Ind. Appl.* **2019**, *55*, 4225–4234. [CrossRef]
- 118. Rauf, H.; Khalid, M.; Arshad, N. Machine learning in state of health and remaining useful life estimation: Theoretical and technological development in battery degradation modelling. *Renew. Sustain. Energy Rev.* **2022**, *156*, 111903. [CrossRef]
- 119. Liu, Y.; He, Y.; Bian, H.; Guo, W.; Zhang, X. A review of lithium-ion battery state of charge estimation based on deep learning: Directions for improvement and future trends. *J. Energy Storage* **2022**, *52*, 104664. [CrossRef]
- Chemali, E.; Kollmeyer, P.J.; Preindl, M.; Emadi, A. State-of-charge estimation of Li-ion batteries using deep neural networks: A machine learning approach. J. Power Sources 2018, 400, 242–255. [CrossRef]
- 121. Cunningham, P.; Cord, M.; Delany, S.J. Supervised learning. In *Machine Learning Techniques for Multimedia: Case Studies on Organization and Retrieval*; Springer: Berlin/Heidelberg, Germany, 2008; pp. 21–49.

- Zhang, C.; Zhang, P.; Wang, Y.; Zhang, L.; Hu, J.; Zhang, W. Support vector machine based lithium-ion battery electrolyte leakage fault diagnosis method. In Proceedings of the 2022 4th International Conference on Smart Power & Internet Energy Systems (SPIES), Beijing, China, 9–12 December 2022; pp. 1880–1886.
- 123. Li, Y. Deep reinforcement learning: An overview. arXiv 2017, arXiv:1701.07274.
- 124. Zhang, K.; Yang, Z.; Başar, T. Multi-agent reinforcement learning: A selective overview of theories and algorithms. In *Handbook of Reinforcement Learning and Control*; Springer: Berlin/Heidelberg, Germany, 2021; pp. 321–384.
- 125. Cui, Z.; Wang, L.; Li, Q.; Wang, K. A comprehensive review on the state of charge estimation for lithium-ion battery based on neural network. *Int. J. Energy Res.* 2022, *46*, 5423–5440. [CrossRef]
- 126. Chen, Z.; Mi, C.C.; Fu, Y.; Xu, J.; Gong, X. Online battery state of health estimation based on Genetic Algorithm for electric and hybrid vehicle applications. *J. Power Sources* **2013**, 240, 184–192. [CrossRef]
- 127. Miguel, E.; Plett, G.L.; Trimboli, M.S.; Oca, L.; Iraola, U.; Bekaert, E. Review of computational parameter estimation methods for electrochemical models. *J. Energy Storage* **2021**, *44*, 103388. [CrossRef]
- 128. Dargan, S.; Kumar, M.; Ayyagari, M.R.; Kumar, G. A survey of deep learning and its applications: A new paradigm to machine learning. *Arch. Comput. Methods Eng.* **2020**, *27*, 1071–1092. [CrossRef]
- 129. Saputri, T.R.D.; Lee, S.W. The application of machine learning in self-adaptive systems: A systematic literature review. *IEEE Access* **2020**, *8*, 205948–205967. [CrossRef]
- 130. Oh, S.; Kim, J.; Moon, I. Hybrid data-driven deep learning model for state of charge estimation of Li-ion battery in an electric vehicle. *J. Energy Storage* 2024, *97*, 112887. [CrossRef]
- 131. Abiodun, O.I.; Jantan, A.; Omolara, A.E.; Dada, K.V.; Mohamed, N.A.; Arshad, H. State-of-the-art in artificial neural network applications: A survey. *Heliyon* 2018, 4, e00938. [CrossRef]
- 132. Feng, F.; Teng, S.; Liu, K.; Xie, J.; Xie, Y.; Liu, B.; Li, K. Co-estimation of lithium-ion battery state of charge and state of temperature based on a hybrid electrochemical-thermal-neural-network model. J. Power Sources 2020, 455, 227935. [CrossRef]
- 133. Vidal, C.; Haußmann, M.; Barroso, D.; Shamsabadi, P.M.; Biswas, A.; Chemali, E.; Ahmed, R.; Emadi, A. Hybrid energy storage system state-of-charge estimation using artificial neural network for micro-hybrid applications. In Proceedings of the 2018 IEEE Transportation Electrification Conference and Expo (ITEC), Long Beach, CA, USA, 13–15 June 2018; pp. 1075–1081.
- 134. Zahid, T.; Xu, K.; Li, W.; Li, C.; Li, H. State of charge estimation for electric vehicle power battery using advanced machine learning algorithm under diversified drive cycles. *Energy* **2018**, *162*, 871–882. [CrossRef]
- 135. Khalid, A.; Sundararajan, A.; Acharya, I.; Sarwat, A.I. Prediction of li-ion battery state of charge using multilayer perceptron and long short-term memory models. In Proceedings of the 2019 IEEE Transportation Electrification Conference and Expo (ITEC), Detroit, MI, USA, 19–21 June 2019; pp. 1–6.
- 136. Ma, L.; Hu, C.; Cheng, F. State of charge and state of energy estimation for lithium-ion batteries based on a long short-term memory neural network. *J. Energy Storage* **2021**, *37*, 102440. [CrossRef]
- Lipu, M.H.; Hannan, M.; Hussain, A.; Ayob, A.; Saad, M.H.; Karim, T.F.; How, D.N. Data-driven state of charge estimation of lithium-ion batteries: Algorithms, implementation factors, limitations and future trends. *J. Clean. Prod.* 2020, 277, 124110. [CrossRef]
- 138. Zhao, F.; Guo, Y.; Chen, B. A review of lithium-ion battery state of charge estimation methods based on machine learning. *World Electr. Veh. J.* 2024, 15, 131. [CrossRef]
- 139. Chen, J.; Ouyang, Q.; Xu, C.; Su, H. Neural network-based state of charge observer design for lithium-ion batteries. *IEEE Trans. Control. Syst. Technol.* **2017**, *26*, 313–320. [CrossRef]
- 140. Yang, F.; Li, W.; Li, C.; Miao, Q. State-of-charge estimation of lithium-ion batteries based on gated recurrent neural network. *Energy* **2019**, *175*, 66–75. [CrossRef]
- 141. Huang, Z.; Yang, F.; Xu, F.; Song, X.; Tsui, K.L. Convolutional gated recurrent unit–recurrent neural network for state-of-charge estimation of lithium-ion batteries. *IEEE Access* 2019, *7*, 93139–93149. [CrossRef]
- 142. Li, A.G. State Estimation in Lithium-Ion Batteries Using Pulse Perturbation and Feedforward Neural Networks. Ph.D. Thesis, Columbia University, New York, NY, USA, 2020.
- 143. Vidal, C.; Kollmeyer, P.; Naguib, M.; Malysz, P.; Gross, O.; Emadi, A. Robust xev battery state-of-charge estimator design using a feedforward deep neural network. *SAE Int. J. Adv. Curr. Pract. Mobil.* **2020**, *2*, 2872–2880. [CrossRef]
- 144. Chen, J.; Zhang, Y.; Wu, J.; Cheng, W.; Zhu, Q. SOC estimation for lithium-ion battery using the LSTM-RNN with extended input and constrained output. *Energy* **2023**, 262, 125375. [CrossRef]
- Li, Y.; Li, K.; Liu, X.; Zhang, L. Fast battery capacity estimation using convolutional neural networks. *Trans. Inst. Meas. Control.* 2020, 0142331220966425. [CrossRef]
- 146. Chen, D.; Zheng, X.; Chen, C.; Zhao, W. Remaining useful life prediction of the lithium-ion battery based on CNN-LSTM fusion model and grey relational analysis. *Electron. Res. Arch.* **2023**, *31*, 633–655. [CrossRef]
- 147. Xu, F.; Yang, F.; Fei, Z.; Huang, Z.; Tsui, K.L. Life prediction of lithium-ion batteries based on stacked denoising autoencoders. *Reliab. Eng. Syst. Saf.* **2021**, 208, 107396. [CrossRef]
- 148. Zhang, G.; Xia, B.; Wang, J.; Ye, B.; Chen, Y.; Yu, Z.; Li, Y. Intelligent state of charge estimation of battery pack based on particle swarm optimization algorithm improved radical basis function neural network. *J. Energy Storage* **2022**, *50*, 104211. [CrossRef]
- 149. Chen, L.; Ding, Y.; Wang, H.; Wang, Y.; Liu, B.; Wu, S.; Li, H.; Pan, H. Online estimating state of health of lithium-ion batteries using hierarchical extreme learning machine. *IEEE Trans. Transp. Electrif.* **2021**, *8*, 965–975. [CrossRef]

- 150. Zhang, S.; Guo, X.; Zhang, X. An improved adaptive unscented kalman filtering for state of charge online estimation of lithium-ion battery. *J. Energy Storage* 2020, *32*, 101980. [CrossRef]
- 151. Shrivastava, P.; Soon, T.K.; Idris, M.Y.I.B.; Mekhilef, S. Overview of model-based online state-of-charge estimation using Kalman filter family for lithium-ion batteries. *Renew. Sustain. Energy Rev.* **2019**, *113*, 109233. [CrossRef]
- 152. Tian, Y.; Lai, R.; Li, X.; Xiang, L.; Tian, J. A combined method for state-of-charge estimation for lithium-ion batteries using a long short-term memory network and an adaptive cubature Kalman filter. *Appl. Energy* **2020**, *265*, 114789. [CrossRef]
- 153. Peng, J.; Luo, J.; He, H.; Lu, B. An improved state of charge estimation method based on cubature Kalman filter for lithium-ion batteries. *Appl. Energy* **2019**, 253, 113520. [CrossRef]
- 154. He, Z.; Yang, Z.; Cui, X.; Li, E. A method of state-of-charge estimation for EV power lithium-ion battery using a novel adaptive extended Kalman filter. *IEEE Trans. Veh. Technol.* **2020**, *69*, 14618–14630. [CrossRef]
- 155. Ling, L.; Wei, Y. State-of-charge and state-of-health estimation for lithium-ion batteries based on dual fractional-order extended Kalman filter and online parameter identification. *IEEE Access* **2021**, *9*, 47588–47602. [CrossRef]
- 156. Bi, Y.; Choe, S.Y. An adaptive sigma-point Kalman filter with state equality constraints for online state-of-charge estimation of a Li (NiMnCo) O₂/Carbon battery using a reduced-order electrochemical model. *Appl. Energy* **2020**, 258, 113925. [CrossRef]
- 157. Cao, Y.; Li, Y.; Zhang, G.; Jermsittiparsert, K.; Nasseri, M. An efficient terminal voltage control for PEMFC based on an improved version of whale optimization algorithm. *Energy Rep.* **2020**, *6*, 530–542. [CrossRef]
- 158. Sankhala, D.; Pali, M.; Lin, K.C.; Jagannath, B.; Muthukumar, S.; Prasad, S. Analysis of bio-electro-chemical signals from passive sweat-based wearable electro-impedance spectroscopy (EIS) towards assessing blood glucose modulations. *arXiv* 2021, arXiv:2104.01793
- 159. He, L.; Guo, D. An improved coulomb counting approach based on numerical iteration for SOC estimation with real-time error correction ability. *IEEE Access* 2019, 7, 74274–74282. [CrossRef]
- 160. Chen, C.; Xiong, R.; Yang, R.; Shen, W.; Sun, F. State-of-charge estimation of lithium-ion battery using an improved neural network model and extended Kalman filter. *J. Clean. Prod.* **2019**, 234, 1153–1164. [CrossRef]
- 161. Liu, K.; Ashwin, T.; Hu, X.; Lucu, M.; Widanage, W.D. An evaluation study of different modelling techniques for calendar ageing prediction of lithium-ion batteries. *Renew. Sustain. Energy Rev.* **2020**, *131*, 110017. [CrossRef]
- 162. Tian, J.; Xiong, R.; Shen, W.; Lu, J. State-of-charge estimation of LiFePO4 batteries in electric vehicles: A deep-learning enabled approach. *Appl. Energy* 2021, 291, 116812. [CrossRef]
- Ayodele, T.; Ogunjuyigbe, A.; Oyelowo, N. Hybridisation of battery/flywheel energy storage system to improve ageing of lead-acid batteries in PV-powered applications. *Int. J. Sustain. Eng.* 2020, 13, 337–359. [CrossRef]
- 164. Soto, A.; Berrueta, A.; Mateos, M.; Sanchis, P.; Ursúa, A. Impact of micro-cycles on the lifetime of lithium-ion batteries: An experimental study. *J. Energy Storage* **2022**, *55*, 105343. [CrossRef]
- 165. Xu, Y.; Zhang, H.; Yang, F.; Tong, L.; Yan, D.; Yang, Y.; Ren, J.; Ma, L.; Wang, Y. State of charge estimation of supercapacitors based on multi-innovation unscented Kalman filter under a wide temperature range. *Int. J. Energy Res.* 2022, 46, 16716–16735. [CrossRef]
- Cui, Z.; Kang, L.; Li, L.; Wang, L.; Wang, K. A combined state-of-charge estimation method for lithium-ion battery using an improved BGRU network and UKF. *Energy* 2022, 259, 124933. [CrossRef]
- 167. Kumar, P.S.; Kamath, R.N.; Boyapati, P.; Josephson, P.J.; Natrayan, L.; Shadrach, F.D. IoT battery management system in electric vehicle based on LR parameter estimation and ORMeshNet gateway topology. *Sustain. Energy Technol. Assess.* 2022, 53, 102696.
- Liu, X.; Chen, Z.; Zhang, C.; Wu, J. A novel temperature-compensated model for power Li-ion batteries with dual-particle-filter state of charge estimation. *Appl. Energy* 2014, 123, 263–272. [CrossRef]
- 169. Tran, M.K.; Mathew, M.; Janhunen, S.; Panchal, S.; Raahemifar, K.; Fraser, R.; Fowler, M. A comprehensive equivalent circuit model for lithium-ion batteries, incorporating the effects of state of health, state of charge, and temperature on model parameters. *J. Energy Storage* 2021, 43, 103252. [CrossRef]
- 170. Ng, M.F.; Zhao, J.; Yan, Q.; Conduit, G.J.; Seh, Z.W. Predicting the state of charge and health of batteries using data-driven machine learning. *Nat. Mach. Intell.* **2020**, *2*, 161–170. [CrossRef]
- 171. Lv, H.; Huang, X.; Liu, Y. Analysis on pulse charging–discharging strategies for improving capacity retention rates of lithium-ion batteries. *Ionics* 2020, *26*, 1749–1770. [CrossRef]
- 172. Wang, X.; Wei, X.; Dai, H. Estimation of state of health of lithium-ion batteries based on charge transfer resistance considering different temperature and state of charge. *J. Energy Storage* **2019**, *21*, 618–631. [CrossRef]
- 173. Wu, T.; Wang, C.; Hu, Y.; Liang, Z.; Fan, C. Research on electrochemical characteristics and heat generating properties of power battery based on multi-time scales. *Energy* **2023**, *265*, 126416. [CrossRef]
- 174. Vishnu, C.; Saleem, A. Adaptive Integral Correction-Based State of Charge Estimation Strategy for Lithium-Ion Cells. *IEEE Access* 2022, 10, 69499–69510. [CrossRef]
- 175. Li, X.; Wang, Z.; Zhang, L. Co-estimation of capacity and state-of-charge for lithium-ion batteries in electric vehicles. *Energy* **2019**, 174, 33–44. [CrossRef]
- 176. Song, X.; Yang, F.; Wang, D.; Tsui, K.L. Combined CNN-LSTM network for state-of-charge estimation of lithium-ion batteries. *IEEE Access* **2019**, *7*, 88894–88902. [CrossRef]
- 177. Wang, Y.; Chen, Z. A framework for state-of-charge and remaining discharge time prediction using unscented particle filter. *Appl. Energy* **2020**, *260*, 114324. [CrossRef]

- 178. Liu, B.; Tang, X.; Gao, F. Joint estimation of battery state-of-charge and state-of-health based on a simplified pseudo-twodimensional model. *Electrochim. Acta* 2020, 344, 136098. [CrossRef]
- 179. Plett, G.L. Extended Kalman filtering for battery management systems of LiPB-based HEV battery packs: Part 3. State and parameter estimation. *J. Power Sources* **2004**, 134, 277–292. [CrossRef]
- 180. Xiong, R.; He, H.; Sun, F.; Zhao, K. Evaluation on state of charge estimation of batteries with adaptive extended Kalman filter by experiment approach. *IEEE Trans. Veh. Technol.* **2012**, *62*, 108–117. [CrossRef]
- 181. Sun, F.; Hu, X.; Zou, Y.; Li, S. Adaptive unscented Kalman filtering for state of charge estimation of a lithium-ion battery for electric vehicles. *Energy* **2011**, *36*, 3531–3540. [CrossRef]
- 182. Lee, S.; Kim, J.; Lee, J.; Cho, B.H. The state and parameter estimation of an Li-ion battery using a new OCV-SOC concept. In Proceedings of the 2007 IEEE Power Electronics Specialists Conference, Orlando, FL, USA, 17–21 June 2007; pp. 2799–2803.
- Xing, Y.; He, W.; Pecht, M.; Tsui, K.L. State of charge estimation of lithium-ion batteries using the open-circuit voltage at various ambient temperatures. *Appl. Energy* 2014, 113, 106–115. [CrossRef]
- 184. Lu, L.; Han, X.; Li, J.; Hua, J.; Ouyang, M. A review on the key issues for lithium-ion battery management in electric vehicles. *J. Power Sources* **2013**, 226, 272–288. [CrossRef]
- 185. Hansen, T.; Wang, C.J. Support vector based battery state of charge estimator. J. Power Sources 2005, 141, 351–358. [CrossRef]
- Hossain, M.; Haque, M.; Arif, M.T. Kalman filtering techniques for the online model parameters and state of charge estimation of the Li-ion batteries: A comparative analysis. *J. Energy Storage* 2022, *51*, 104174. [CrossRef]
- 187. Awadallah, M.A.; Venkatesh, B. Accuracy improvement of SOC estimation in lithium-ion batteries. *J. Energy Storage* 2016, *6*, 95–104. [CrossRef]
- 188. Andre, D.; Appel, C.; Soczka-Guth, T.; Sauer, D.U. Advanced mathematical methods of SOC and SOH estimation for lithium-ion batteries. *J. Power Sources* 2013, 224, 20–27. [CrossRef]
- 189. Huang, S.C.; Tseng, K.H.; Liang, J.W.; Chang, C.L.; Pecht, M.G. An online SOC and SOH estimation model for lithium-ion batteries. *Energies* 2017, *10*, 512. [CrossRef]
- 190. Xile, D.; Caiping, Z.; Jiuchun, J. Evaluation of SOC estimation method based on EKF/AEKF under noise interference. *Energy Procedia* **2018**, *152*, 520–525. [CrossRef]
- 191. Grewal, M.S.; Andrews, A.P. Kalman Filtering: Theory and Practice with MATLAB; John Wiley & Sons: Hoboken, NJ, USA, 2014.
- 192. Chemali, E.; Kollmeyer, P.J.; Preindl, M.; Ahmed, R.; Emadi, A. Long short-term memory networks for accurate state-of-charge estimation of Li-ion batteries. *IEEE Trans. Ind. Electron.* 2017, 65, 6730–6739. [CrossRef]
- 193. Zheng, L.; Zhu, J.; Wang, G.; Lu, D.D.C.; He, T. Differential voltage analysis based state of charge estimation methods for lithium-ion batteries using extended Kalman filter and particle filter. *Energy* **2018**, *158*, 1028–1037. [CrossRef]
- Zhang, S.; Xie, C.; Zeng, C.; Quan, S. SOC estimation optimization method based on parameter modified particle Kalman Filter algorithm. *Clust. Comput.* 2019, 22, 6009–6018. [CrossRef]
- 195. Dai, K.; Wang, J.; He, H. An improved SOC estimator using time-varying discrete sliding mode observer. *IEEE Access* 2019, 7, 115463–115472. [CrossRef]
- 196. Sethia, G.; Nayak, S.K.; Majhi, S. An approach to estimate lithium-ion battery state of charge based on adaptive Lyapunov super twisting observer. *IEEE Trans. Circuits Syst. Regul. Pap.* 2020, *68*, 1319–1329. [CrossRef]
- 197. Nguyen, Q.D.; Huang, S.C. Synthetic adaptive fuzzy disturbance observer and sliding-mode control for chaos-based secure communication systems. *IEEE Access* 2021, *9*, 23907–23928.
- Chen, J.; Xu, G.; Zhou, Z. Data-driven learning-based Model Predictive Control for energy-intensive systems. *Adv. Eng. Inform.* 2023, 58, 102208. [CrossRef]
- 199. Chen, E.X. Multi-Objective Building System Control Optimization Using Machine-Learning-Based Techniques. Ph.D. Thesis, Harvard University, Cambridge, MA, USA, 2023.
- Xie, Y.; Wang, C.; Hu, X.; Lin, X.; Zhang, Y.; Li, W. An MPC-based control strategy for electric vehicle battery cooling considering energy saving and battery lifespan. *IEEE Trans. Veh. Technol.* 2020, 69, 14657–14673. [CrossRef]
- 201. Cui, Z.; Dai, J.; Sun, J.; Li, D.; Wang, L.; Wang, K. Hybrid methods using neural network and Kalman filter for the state of charge estimation of lithium-ion battery. *Math. Probl. Eng.* 2022, 2022, 9616124. [CrossRef]
- Xing, L.; Ling, L.; Gong, B.; Zhang, M. State-of-charge estimation for Lithium-Ion batteries using Kalman filters based on fractional-order models. *Connect. Sci.* 2022, 34, 162–184. [CrossRef]
- Chen, J.; Zhang, Y.; Li, W.; Cheng, W.; Zhu, Q. State of charge estimation for lithium-ion batteries using gated recurrent unit recurrent neural network and adaptive Kalman filter. J. Energy Storage 2022, 55, 105396. [CrossRef]
- 204. Fahmy, H.M.; Swief, R.A.; Hasanien, H.M.; Alharbi, M.; Maldonado, J.L.; Jurado, F. Hybrid State of Charge Estimation of Lithium-Ion Battery Using the Coulomb Counting Method and an Adaptive Unscented Kalman Filter. *Energies* 2023, 16, 5558. [CrossRef]
- Li, Y.; Ye, M.; Wang, Q.; Lian, G.; Xia, B. An Improved Model Combining Machine Learning and Kalman Filtering Architecture for State of Charge Estimation of Lithium-Ion Batteries. *Green Energy Intell. Transp.* 2024, 3, 100163. [CrossRef]
- 206. Das, K.; Kumar, R. Electric vehicle battery capacity degradation and health estimation using machine-learning techniques: A review. *Clean Energy* **2023**, *7*, 1268–1281. [CrossRef]

- 207. Lee, S.B.; Thiagarajan, R.S.; Subramanian, V.R.; Onori, S. Advanced Battery Management Systems: Modeling and Numerical Simulation for Control. In Proceedings of the 2022 American Control Conference (ACC), Atlanta, GA, USA, 8–10 June 2022; pp. 4403–4414.
- 208. El-Dalahmeh, M. Capacity Estimation and Trajectory Prediction of Lithium-ion Batteries Based on Time-Frequency Analysis and Machine Learning Algorithms. Ph.D. Thesis, Teesside University, Middlesbrough, UK, 2023.
- 209. Lavin, A.; Krakauer, D.; Zenil, H.; Gottschlich, J.; Mattson, T.; Brehmer, J.; Anandkumar, A.; Choudry, S.; Rocki, K.; Baydin, A.G.; et al. Simulation intelligence: Towards a new generation of scientific methods. *arXiv* 2021, arXiv:2112.03235.
- Hesse, H.C.; Schimpe, M.; Kucevic, D.; Jossen, A. Lithium-ion battery storage for the grid—A review of stationary battery storage system design tailored for applications in modern power grids. *Energies* 2017, 10, 2107. [CrossRef]
- Connolly, D.; Lund, H.; Mathiesen, B.V.; Leahy, M. A review of computer tools for analysing the integration of renewable energy into various energy systems. *Appl. Energy* 2010, 87, 1059–1082. [CrossRef]
- 212. Openmod. Open Energy Modelling Initiative. Available online: https://openmod-initiative.org/ (accessed on 26 August 2022).
- 213. Volosencu, C. Introductory Chapter: Matlab and Simulink Applications. In *MATLAB Applications in Engineering;* IntechOpen: London, UK, 2022; p. 3.
- 214. Khan, A.A.; Naeem, M.; Iqbal, M.; Qaisar, S.; Anpalagan, A. A compendium of optimization objectives, constraints, tools and algorithms for energy management in microgrids. *Renew. Sustain. Energy Rev.* **2016**, *58*, 1664–1683. [CrossRef]
- Kanchev, H.; Lazarov, V.; Francois, B. Environmental and economical optimization of microgrid long term operational planning including PV-based active generators. In Proceedings of the 2012 15th International Power Electronics and Motion Control Conference (EPE/PEMC), Novi Sad, Serbia, 4–6 September 2012; pp. LS4b-2.1-1–LS4b-2.1-8.
- Khederzadeh, M. Optimal Automation Level in Microgrids. In Proceedings of the 22nd International Conference and Exhibition on Electricity Distribution (CIRED 2013), Stockholm, Sweden, 10–13 June 2013.
- 217. Reddy, Y.J.; Kumar, Y.P.; Kumar, V.S.; Raju, K.P. Distributed ANNs in a layered architecture for energy management and maintenance scheduling of renewable energy HPS microgrids. In Proceedings of the 2012 International Conference on Advances in Power Conversion and Energy Technologies (APCET), Mylavaram, India, 2–4 August 2012; pp. 1–6.
- 218. The MathWorks, Inc. MATLAB/Simulink. Available online: https://www.mathworks.com/products/simulink.html (accessed on 19 November 2023).
- 219. Surya, S.; Samanta, A.; Marcis, V.; Williamson, S. Smart core and surface temperature estimation techniques for health-conscious lithium-ion battery management systems: A model-to-model comparison. *Energies* **2022**, *15*, 623. [CrossRef]
- Fryza, T.; Svobodova, J.; Adamec, F.; Marsalek, R.; Prokopec, J. Overview of parallel platforms for common high performance computing. *Radioengineering* 2012, 21, 436–444.
- 221. Bistline, J.; Cole, W.; Damato, G.; DeCarolis, J.; Frazier, W.; Linga, V.; Marcy, C.; Namovicz, C.; Podkaminer, K.; Sims, R.; et al. Energy storage in long-term system models: A review of considerations, best practices, and research needs. *Prog. Energy* 2020, 2, 032001. [CrossRef]
- 222. kaun, B. Storage Value Estimation Tool (StorageVET[®]). A Publicly Available, Web-Hosted, Energy Storage Value Simulation Tool. Available online: https://www.storagevet.com/ (accessed on 26 August 2022).
- 223. Energy Toolbase. StorageVET. Available online: https://www.energytoolbase.com (accessed on 8 August 2024).
- 224. Nguyen, T.A.; Byrne, R.H. Software tools for energy storage valuation and design. *Curr. Sustain. Energy Rep.* 2021, *8*, 156–163. [CrossRef]
- 225. DER-VET Developer Team | EPRI. Distributed Energy Resource Value Estimation Tool (DER-VET[™]). Available online: https://www.der-vet.com/ (accessed on 26 August 2022).
- 226. EPRI U.G. Technical Documentation for the Distributed Energy Resources Value Estimation Tool (DERVET TM) V0. 1.1 Technical Update; EPRI: Palo Alto, CA, USA, 2020.
- 227. EPRI. Distributed Energy Resource Value Estimation Tool (DER-VET). Available online: https://www.epri.com (accessed on 8 August 2024).
- 228. Gemmer, R.; Paulsen, K. HOMER. Technical Report; EERE Publication and Product Library: Washington, DC, USA, 2022.
- 229. HOMER Energy LLC. Hybrid Optimization of Multiple Energy Resources. Available online: https://www.homerenergy.com/ (accessed on 26 August 2022).
- 230. Lambert, T.; Gilman, P.; Lilienthal, P. Micropower system modeling with HOMER. Integr. Altern. Sources Energy 2006, 1, 379–385.
- 231. HOMER Energy. Distributed Energy Resource Value Estimation Tool (DER-VET). Available online: https://www.homerenergy. com (accessed on 8 August 2024).
- 232. Bahramara, S.; Moghaddam, M.P.; Haghifam, M. Optimal planning of hybrid renewable energy systems using HOMER: A review. *Renew. Sustain. Energy Rev.* **2016**, *62*, 609–620. [CrossRef]
- 233. Weniger, J.; Tjaden, T.; Orth, N.; Maier, S. Performance Simulation Model for PV-Battery Systems (PerMod). 2020. Available online: https://solar.htw-berlin.de/wp-content/uploads/HTW-PerMod-Dokumentation.pdf (accessed on 26 August 2022).
- 234. PerModAC.com. PerModAC. Available online: https://www.permodac.com (accessed on 15 December 2023).
 235. Woniger L: Tiadon T: Orth N: Major S. *Parformance Simulation Model for PV Pattern Systems (ParMod)*: University of
- 235. Weniger, J.; Tjaden, T.; Orth, N.; Maier, S. *Performance Simulation Model for PV-Battery Systems (PerMod)*; University of Applied Sciences Berlin (HTW Berlin): Berlin, Germany, 2023.
- 236. Pacific Northwest National Laboratory. GridLAB-D[™]. Available online: https://www.pnnl.gov/available-technologies/gridlabdtm (accessed on 19 September 2022).

- 237. Möller, M.; Kucevic, D.; Collath, N.; Parlikar, A.; Dotzauer, P.; Tepe, B.; Englberger, S.; Jossen, A.; Hesse, H. SimSES: A holistic simulation framework for modeling and analyzing stationary energy storage systems. *J. Energy Storage* 2022, 49, 103743. [CrossRef]
- Gao, D.W.; Muljadi, E.; Tian, T.; Miller, M. Software Comparison for Renewable Energy Deployment in a Distribution Network; Technical Report; National Renewable Energy Lab.(NREL): Golden, CO, USA, 2017.
- 239. NREL Transforming Energy. Battery Lifetime Analysis and Simulation Tool (BLAST) Suite. Available online: https://www.nrel. gov/transportation/blast.html (accessed on 26 August 2022).
- 240. NREL Transforming Energy. System Advisor Model (SAM). Available online: https://sam.nrel.gov/ (accessed on 26 August 2023).
- 241. DiOrio, N.; Dobos, A.; Janzou, S.; Nelson, A.; Lundstrom, B. *Technoeconomic Modeling of Battery Energy Storage in SAM*; Technical Report; National Renewable Energy Lab.(NREL): Golden, CO, USA, 2015.
- 242. DiOrio, N.; Denholm, P.; Hobbs, W.B. A model for evaluating the configuration and dispatch of PV plus battery power plants. *Appl. Energy* **2020**, 262, 114465. [CrossRef]
- Smith, K.; Saxon, A.; Keyser, M.; Lundstrom, B.; Cao, Z.; Roc, A. Life prediction model for grid-connected Li-ion battery energy storage system. In Proceedings of the 2017 American Control Conference (ACC), Seattle, WA, USA, 24–26 May 2017; pp. 4062–4068.
- 244. Spotnitz, R. Simulation of capacity fade in lithium-ion batteries. J. Power Sources 2003, 113, 72–80. [CrossRef]
- 245. DiOrio, N.A. An Overview of the Automated Dispatch Controller Algorithms in the System Advisor Model (SAM); National Renewable Energy Lab.(NREL): Golden, CO, USA, 2017.
- 246. DiOrio, N.A.; Freeman, J.M.; Blair, N. DC-connected solar plus storage modeling and analysis for behind-the-meter systems in the system advisor model. In Proceedings of the 2018 IEEE 7th World Conference on Photovoltaic Energy Conversion (WCPEC)(A Joint Conference of 45th IEEE PVSC, 28th PVSEC & 34th EU PVSEC), Waikoloa, HI, USA, 10–15 June 2018; pp. 3777–3782.
- 247. Blair, N.; DiOrio, N.; Freeman, J.; Gilman, P.; Janzou, S.; Neises, T.; Wagner, M. System Advisor Model (SAM) General Description (Version 2017.9. 5); Technical Report; National Renewable Energy Laboratory (NREL): Golden, CO, USA, 2018.
- Naumann, M.; Truong, C.N.; Schimpe, M.; Kucevic, D.; Jossen, A.; Hesse, H.C. SimSES-Software for Techno-Economic Simulation of Stationary Energy Storage Systems. In Proceedings of the International ETG Congress 2017, Bonn, Germany, 28–29 November 2017; pp. 1–6.
- Kucevic, D.; Tepe, B.; Englberger, S.; Parlikar, A.; Mühlbauer, M.; Bohlen, O.; Jossen, A.; Hesse, H. Standard battery energy storage system profiles: Analysis of various applications for stationary energy storage systems using a holistic simulation framework. *J. Energy Storage* 2020, *28*, 101077. [CrossRef]
- 250. Technical University of Munich. Simulation of Stationary Energy Storage Systems (SimSES). Available online: https://www.tum.de (accessed on 8 August 2024).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.